

ECONOMIC VALUE ADDED ASSESSMENT APPLIED TO THE PORTUGUESE SME DURING THE CRISIS PERIOD

Catarina Gonçalves de Oliveira

Dissertation submitted as partial requirement for the conferral of

Master in Accounting

Supervisor:

Prof. Doutor Pedro Leite Inácio, Assistant Professor, ISCTE Business School Finance Department

April, 2014

Abstract

Performance measurement has, over the past two decades, been a popular topic amongst academic and professional practitioners. Using a database containing the financial statements of the biggest and best 315 Portuguese SME, we assessed the economic value added by those companies for the triennium of 2008-2010. The findings indicate that globally, for the period of 2008-2010, the companies have not created value for their shareholders. Nevertheless, "Electrical Equipment" and "Commercial Services and Supplies" were the only two sectors that contributed, with a positive MVA, to this globally negative scenario. These findings may be useful for financial managers, investors and corporate finance consultants.

JEL classification: M41, G32

Keywords: Performance, EVA, WACC, ROIC.

Resumo

A medição do desempenho tem sido, ao longo das últimas duas décadas, um tema muito investigado pelos académicos e utilizado pelos profissionais de análise e avaliação de empresas. Partindo de uma base de dados das demonstrações financeiras das 315 maiores e melhores PME Portuguesas, no período compreendido entre 2008 e 2010, avaliamos o valor económico acrescentado por essas empresas para o mesmo período. Os resultados indicam que, globalmente, para o período de 2008-2010, as empresas não criaram valor para os seus acionistas. No entanto, os sectores "Material Eléctrico e de Precisão" e "Serviços Comerciais" foram os dois únicos setores que contribuíram com um MVA positivo, apesar deste cenário globalmente negativo. Estes resultados podem ser úteis para os gestores financeiros, investidores e consultores de finanças empresariais.

Classificação JEL: M41, G32

Palavras-chave : Performance, EVA, WACC, ROIC.

To my family and my supervisor, Prof. Doutor Pedro Leite Inácio

Contents

Abstract	ii
Resumo	iii
List of Tables	vi
List of Figures	vii
Chapter 1. INTRODUCTION	1
Chapter 2. LITERATURE REVIEW	5
2.1. EVA: an indicator for value	5
2.2. EVA and MVA relationship	7
2.3. EVA: a multipurpose tool	8
2.4. EVA and accounting adjustments	11
Chapter 3. CHARACTERIZATION OF THE SAMPLE, METHODOLOGY AND EV	ΛA
ACCOUNTING ADJUSTMENTS	13
3.1. Sample Database Description	13
3.2. Methodology	18
3.3. EVA accounting adjustments	29
Chapter 4. PERFORMANCE ASSESSMENT RESULTS	31
4.1. EVA components and EVA Spread analysis	32
4.2. MVA and RMVA analysis	36
4.3. EVA and MVA rankings	39
Chapter 5. CONCLUDING REMARKS	44
References	47

Appendix A. Annexes

List of Tables

3.1	Portrait of the Portuguese SME in 2007	14
3.2	Key indicators by sector (2008-2010)	16
3.3	Average company key indicators by sector (2008-2010)	17
3.4	Descriptive statistics for the key indicators $(2008, 2009 \text{ and } 2010)$	18
3.5	Estimates for the market risk premium	23
3.6	Levered and unlevered beta for the European sectors	25
3.7	Estimates for country risk coefficient	26
3.8	Estimated unlevered betas: Europe and Portugal	27
4.1	Summary of EVA and MVA results	31
4.2	Descriptive statistics for EVA main components (2008, 2009 and 2010) $$	32
4.3	EVA main components by sector $(2008, 2009 \text{ and } 2010)$	34
4.4	Global MVA descriptive statistics	37
4.5	Total MVA, average RMVA, best and worst company (2008-2010)	38
4.6	EVA TOP+	40
4.7	EVA TOP-	42
4.8	TOP RMVA	43
A.1	Sector names and Portuguese correspondents	52
A.2	List of abbreviations and acronyms	53

List of Figures

3.1	Number of companies by sector	15
3.2	Invested Capital	20
4.1	Weighted average EVA Spread, ROIC and WACC (2008, 2009 and 2010) $$	36

CHAPTER 1

INTRODUCTION

Independently of their size, activity or nationality, the primary goal of all companies worldwide is to create value for their shareholders (Van der Poll, Booyse, Pienaar, Büchner and Foot (2011)). Thus, objectives must be established by those shareholders or by their agents (the managers), so as to determine which course of action should be taken by the company's employees in order to achieve value creation. However, the truly relevant question is "How do companies know that they're staying in course? How can managers make sure that their efforts are, indeed, being directed into the creation of value?".

The answer to this question lies in performance measurement. The stakeholders of a company should, at all times, be conscious of its performance in order to plan, evaluate and make decisions for the future (Van der Poll et al. (2011)). It is only through the measurement of a company's performance that its employees, managers, investors and other stakeholders assure that the right course is being taken, in order to fulfill the company's ultimate purpose, which is to create value for its shareholders.

Over the last two decades, the rapid evolution of technology and globalization has brought new and more demanding challenges to today's managers and business executives (Abdeen and Haight (2002)). These changes have affected the nature and scale of several activity sectors, and combined with the financial and economic crisis, that has settled from the years of 2007 and 2008, have unquestionably shook economies and financial systems around the world. The economic and financial crisis contaminated global economies and had serious impacts for companies worldwide, particularly for those that fit the category of SME, which are most vulnerable to changes in market conditions. Besides the drop in demand for products and services, which weakened the position of companies, there is yet another factor that penalizes them, since they now have greater difficulty in financing their activity due to the tightening of the conditions for obtaining credit (OECD (2009)).

Because of all the above reasons, the need for control and performance evaluation has been increased and so was the need to find performance measurement tools that allow shareholders to filter through this turbulent environment and rigorously evaluate their companies' performance.

In order to better understand the effect of this crisis in Portugal, it is important to characterize the Portuguese business environment, represented in more than 95% by small and medium enterprises (Instituto Nacional de Estatística (2010)).

The category of micro, small and medium enterprises (SME), as defined by the Commission Recommendation of May 6th, 2003, in the Official Journal of the European Union, is the group of companies which employ fewer than 250 persons and whose annual turnover does not exceed 50 million euro or annual balance sheet total assets not exceeding 43 million euro. According to data from Instituto Nacional de Estatística (INE) and Instituto de Apoio às Pequenas e Médias Empresas e à Inovação (IAPMEI), more than 99% of the Portuguese companies fit in this definition, assuming a fundamental role in our economy. In 2007, these companies ensured over 3 million jobs and have generated a total revenues amount of approximately 250 thousand million euro.

Thus, it becomes relevant to investigate the behavior of these companies under the beginning of the crisis period, leading us to the two central questions in this dissertation: "Did the biggest and best Portuguese SME create value during early years of the international crisis period?" and "Which sectors have contributed most to the creation of value in this period?".

In Portugal, the effects of this crisis are manifested by a permanent decline in GDP, as described in the annual report by Banco de Portugal (2013, p. 27): "In the last quarter of 2012, GDP stood 7 per cent below the level observed in end 2010 (and more than 8 per cent below the level in 2008Q1). This accumulated decline in Portuguese GDP exceeds by far the average magnitude of recessions in advanced economies. This episode has now lasted for 9 quarters, which also exceeds the average duration of historical recessions in advanced economies, whatever its cause or level of synchronization.". A change seems to be needed in order to gain a strong competitive presence in the global markets.

Foltin (1999, p. 41) stated that "to remain competitive, it is essential for these companies to be meticulous in maintaining performance". So, the question seems to be "how can we measure and maintain or improve corporate performance?" Author Ismail (2013) helps us answer this question, stating that "Company performance can be measured by using various techniques. Company performance measurement can be a quantitative or qualitative characterization of performance. Qualitative or non-financial measures such as internal coordination, the innovation process and brand image are said to be some of the most important qualitative performance factors of the company. These measures refer to the company's overall capability in producing quality activities, in a way that may lead to improvement in business performance. Quantitative performance refers to physical measurement that enables investors to evaluate business activities through financial statements of the company" (Ismail (2013, p.1757)).

However, performance is not easy to measure and attempting to do so may constitute a problem since some resistance takes place either by the department heads or even from the employees themselves (Foltin (1999)). Given that compensation systems should be closely linked to how the managers perform their functions, the topic of performance evaluation is, above all, very sensitive and controversial (Singh and Mehta (2012)).

The more complex the business, the greater the subjectivity and imprecision of the performance measure adopted and, if it is not the most appropriate measure or it is not correctly applied, it can lead the managers to make the wrong decisions and destroy value for the company, instead of creating it (Singh and Mehta (2012)).

Until the early 90s, many managers used performance measures that are nowadays considered more traditional and conservative, such as earnings per share (EPS), return on equity (ROE), operating profit (OP) and net income (NI). Although these measures are easy to apply, they might not reflect, in an accurate way, the economic reality of a company. Despite having been able to satisfy the needs of companies from the industrial era, these measures can no longer keep up with all the innovations and new needs identified in today's business environment (Ismail (2013)). Moreover, these measures make it difficult to establish a direct comparison of the performance between companies from different sectors or different countries.

A new tool for measuring performance stands out from the rest. Created by Stern Stewart & Co., Economic Value Added (EVA) has two fundamental purposes for a company that wants to achieve prosperity: not only serves as a tool for measuring performance, but can also be used as a tool for incentive compensation. (Kleiman (1999)). EVA is described as a "framework for a complete financial management and incentive compensation system that can guide every decision the company makes, from the boardroom to the shop floor" (Stewart III and Ehrbar (1999, p.18)).

This tool has revolutionized the vision of value creation, which is the ultimate goal for shareholders and managers (McConville (1994)). Companies like Harnischfeger Industries, Manitowoc, Siemens AG, and Coca-Cola Company have embraced EVA as a tool for performance measurement or as compensation plan (Binnersley (1996)).

The remainder of this dissertation is organized as follows: chapter 2 presents the literature review. Chapter 3 outlines the description and characterization of the sample database and the main methodological features of EVA computation. Chapter 4 reports the empirical results. Finally, chapter 5 summarizes the concluding remarks.

CHAPTER 2

LITERATURE REVIEW

The present chapter will focus on the review of the main topics published regarding EVA. First, section 2.1 reviews the works on EVA as an indicator for value. Section 2.2 brings insight from previous studies on the relationship between EVA and MVA. Section 2.3 presents the literature on other important roles that EVA assumes in management. Finally, section 2.4 focuses on the review of the main publications on the adjustments needed in order to apply EVA as a valuation tool.

2.1. EVA: an indicator for value

According to Ray (2001), value is defined as the quality, or the price, which is perceived, or paid, by the customer. Therefore, the continuous increase in value for an organization is attained by the creation of wealth for its shareholders, by satisfying the needs and expectations of customers, suppliers, employees and other stakeholders. Ray (2001) developed a theoretical analysis that offers a new definition of value and suggests that there is a missing link in the EVA process, which is productivity. In his work, he reviewed a series of theories and evidence regarding EVA, having placed this model within the larger context of valuation metrics and concluded that the use of EVA allows managers to concentrate their attention on the most productive areas of the company.

It seems logical that when a company's share price increases, it has created value while, on the other hand, if a company destroys value, its share prices should register a decrease. According to Lovata and Costigan (2002), the primary goal of a company's managers should be to increase shareholder value. Stewart (1994) stated that EVA translates shareholders' wealth directly, in a way that traditional accounting measures cannot. The author also noted that, in theory, shareholders' wealth is maximized only when the company's net present value, and hence the present value of EVA, is maximized. Thus, EVA might just be a good estimator for the true value of a company.

Raiyani and Joshi (2011) have published a case study on the use of EVA as a performance measure in which they suggested that the ultimate goal of EVA is to understand if the cost of capital being employed on the expansion and operation of a business does or does not create real value for its shareholders. Moreover, the authors stated that, in order to increase shareholder value, EVA should be used as a multidimensional tool and not just as a performance measure. By doing this, EVA becomes more than just a financial measure: it becomes a tool for value creation.

Many authors have tried to establish a relation between accounting numbers and stock returns, but have forgotten that wealth maximization is actually more than just maximizing a company's return rates. For instance, Athanassakos (2007) noted that companies with better stock market performances are more likely to use EVA; Sharma and Kumar (2010) reinforced this idea by stating that companies that implement EVA seem to present higher profitability than their competitors. A survey conducted by Grant (1996) concluded that the use of EVA strongly impacts the company's value, confirming the existence of a relationship between EVA and firm value. Moreover, Wallace (1994) supported Grant's findings when examining the changes made by EVA adopters and concluded that the companies that implement EVA dispose of more assets and, hence, need fewer new investments, resulting in the creation of more value without the need of further capital investment.

Weaver (2001) conducted a survey with the purpose of detailing how the EVA adopting companies measured EVA. In accordance with Stewart (1991), Weaver concluded that companies adopt EVA because they believe that it is the tool most closely linked to stock prices and economic analysis. By pursuing a target EVA, managers expect to increase the price of their company's shares and, ultimately, increase the company's overall market value. Bao and Bao (1998) published a study on the usefulness of economic value added and abnormal earnings using a sample of US firms. The results suggest that EVA is a better explanatory factor for market returns than other traditional accounting measures. The findings of Worthington and West (2004) supported these same conclusions.

Authors such as Chen and Dodd (1997) argued that even though EVA provides more information on stock association than traditional accounting measures, it should not replace them, since the best results on market value estimation come from their complementary use.

On the contrary perspective, some authors concluded that EVA does not perform this well when predicting a firm's value. Instead of being superior to all other financial measures, there are claims that EVA is not superior to other accounting measures as value indicators. In fact, some authors argue, for instance, that earnings are better predictors of a firm's value rather than EVA—see, for example, Erasmus (2008), Tham (2001), Cordeiro and Kent (2001), Peixoto (2001), Biddle, Bowen and Wallace (1997).

2.2. EVA and MVA relationship

Stewart (1991, p. 153) advocated that "EVA ties directly to the intrinsic market value of any company". EVA is designed to subtract the capital charges (of existing and new investments) from operating profits in each year. According to Stewart (1991), by projecting and discounting EVAs to present value, we get the market value created by management using the capital employed in the company, to which we call Market Value Added (MVA).

Many studies have been made with the purpose of studying the relationship between EVA and MVA and authors have reached different results. In fact, MVA is simply the difference between the implied enterprise value, in share market prices, and the company's invested capital (usually measured by its book value) and it should reflect the present value of its future expected EVAs. When a company's EVA increases, its market value added also increases, reflecting the creation of value by the management's actions. MVA reflects the premium (or discount) of the market prices relative to the total capital invested in the company.

Stewart (1991) analyzed the correlation between several indicators of a company's performance and MVA and concluded that EVA is the indicator that better correlates with the MVA of a company. Finegan (1991) and O'Byrne (1996) stated that EVA is the one measure that is systematically bound to a company's market value and also that it is also a tool that allows the understanding of investor's expectations. Authors such as Lehn and Makhija (1997), Clinton and Chen (1998), Herzberg (1998) and Elali (2007) supported the conclusion that EVA has a high correlation with MVA and stock prices.

When investigating EVA's predictive power in explaining MVA, some authors observed that EVA is the indicator that best correlates to MVA when compared to traditional indicators such as return on assets, return on capital employed or earnings per share—for more details, please see Uyemura, Kantor and Petit (1996), O'Byrne (1996), McCormack and Vytheeswaran (1998). Case studies on Automobile and Fertilizer industries in India, were developed by Ghanbari and More (2007) and Joshi (2011) on the relationship between EVA and MVA, which supported Stewart's findings.

However, there are authors that argued that EVA is not the measure with the highest correlation with MVA, pointing other measures that present better explanatory power when compared to EVA. Amongst these authors is Fernandez (2001), who found that NOPAT had a better correlation with MVA in 296 of 582 US companies, and for 210 of those companies that the correlation between EVA and MVA was actually negative. Kramer and Pushner (1997) supported Fernandez's findings when studying the strength of the relationship between EVA and MVA.

DeWet (2005) studied the strength of the relationship between MVA and EVA using data from South African listed companies. The results revealed that, for the period between 1994 and 2004 and on a year-on-year basis, EVA did not show the strongest correlation with MVA. The author found a stronger relationship between MVA and cash flow from operations.

2.3. EVA: a multipurpose tool

Even though most of the work published on EVA focuses on its use as a tool for assessing the creation (or destruction) of shareholder wealth, EVA stands out as a management tool because it can be applied in several areas that concern a company's management. In this section we review some of the work published on EVA as a tool for decision making and incentive compensation.

EVA as a decision orienting tool. Capital budgeting is the process of evaluating a company's investments and allocating its resources. The decisions made by capital budgeting are normally based on the cash flows associated with the investment (or resource) upon which a company may undertake. However, EVA may be used as a decision making tool concerning new projects and even prospect mergers and acquisitions. Abdeen and Haight (2002) established a comparison between the performance of EVA user companies with non-user Fortune 500 companies, between the years of 1997 and 1998. The authors concluded that the performance of companies using EVA was exceeded by the performance of companies using performance measures such as earnings per share.

Another example of the application of EVA as a decision making tool is given by Abdeen and Haight (2002), who suggest that EVA may be helpful in determining if the acquisition of a company will or will not increase the value of the acquirer company and, therefore, bring additional value to its shareholders.

The expense on advertising, research and development, considered as capital investment according to the EVA philosophy, is another example of the usefulness of EVA in the decision making process—as referred in Dow Theory Forecast (1999), the undertaking of expense in these areas may be based on an EVA assessment. Even the decisions at a product line or individual customer levels can be guided by EVA. By disaggregating the management's information, a company can distinguish the areas where value is being created from those where it is being destroyed. In doing so, managers assure that specific clients or product lines are contributing to the increase of the overall return rates and the satisfaction of shareholders' expectations (Abdeen and Haight (2002)). However, Damodaran (2001) alerted managers to the fact that the simplicity of decision making tools such as EVA or cash flow return on investment, may come at a substantial cost for high growth firms with shifting risk profiles. **EVA as a performance measuring and incentive compensation tool.** In today's business environment, companies must not look past the accurate measurement of their performance (Raymond, St-Pierre and Marchand (2009), Cocca and Alberti (2010)). Companies and their consultants use EVA as a very successful performance measure. In fact, financial theory justifies this metric because it is consistent with valuation principles (earnings, return on investment, market share, cash flow return on investment, etc.) that are important to investors when analyzing the companies' performance.

EVA takes into account the profits generated by a company's resources, without overlooking the cost inherent to those resources. For this reason, companies may use EVA not only as a performance measure, but also as a tool that gathers and interprets several different kinds of financial information, overcoming the problem of the implementation of a complex performance measurement system and its interpretation. Hussein and Laitinen (1998) supported this conclusion after studying the use of several accounting measures in a sample of Finnish service firms.

The performance measurement is also facilitated by the use of only one measure, instead of several individual measures. By using an EVA performance measure, which translates financial and operational indicators in one single language, all the employees may be guided and orient their efforts towards a common goal (Stern, Stewart and Chew (1998)).

Dierks and Patel (1997) demonstrated that the EVA and MVA measures of financial performance can effectively be used in managing a company's operations, in guiding its strategies and in providing incentives to its employees. However, the implementation of value added measures into a company is a timely and costly process, since every individual in the company must understand and be educated in order to make it successful.

With the intention of proposing a performance measurement and management system for SME, Bahri, St-Pierre and Sakka (2011) performed a research based on the analysis of the relationship between the management practices of SME and performance, as measured by EVA. According to the authors, an EVA based performance measurement system provides managers with an assessment on the level of achievement of the company's strategic goals. Bahri et al. (2011) concluded that, even though the impact on the performance indicators is not immediately obvious, the improvements designed to upgrade a company's business practices should not be terminated, since they may only become effective at a later time.

Most companies that implement EVA as a performance measurement tool, also use it as an incentive compensation system as it allows managers to identify which goals are being met and by whom (Abdeen and Haight (2002)).

EVA based bonus plans produce positive results within an organization. One of the major benefits of the use of EVA as an incentive compensation system is that it aligns the interests and objectives of shareholders, managers and first line employees. Wallace (1994) conducted an empirical work with the purpose of comparing the performance of firms that adopted residual income performance measurements (such as EVA) in their compensation plans with the performance of companies that used traditional accounting based incentives. The author concluded that companies that adopted residual income based performance measures had increased their overall performance and overcame some cases of agency conflicts.

Finally, traditional bonus systems do not promote a fair attribution of compensation when distinguishing an employee with a good performance from one whose performance is mediocre see Jensen and Murphy (1990) for a complete in depth statistical analysis of executive compensation for the CEO's salaries and bonuses of 1,400 publicly held companies, from 1974 to 1988.

2.4. EVA and accounting adjustments

Stern Stewart & Co. suggested the application of up to 164 possible adjustments to the EVA computation to achieve the most rigorous translation of a company's value. These adjustments may be sectioned in two categories: the adjustments that are related to the Generally Accepted Accounting Principles (GAAP) and the non-GAAP adjustments, applied to the calculation of NOPAT.

From the wide variety of adjustments possible, companies adopting Stern Stewart & Co. financial system, generally make no more than five to ten adjustments to their published accounts, as emphasized by Stern, Stewart and Chew (1995). Stewart (1991) recommended that the adjustments to the computation of EVA should be in order only when: i) the amounts considered are significant, ii) the adjustments cause a material impact on EVA, iii) employees at all levels can easily grasp the impact made by these adjustments; and iv) the required information is easy to locate. These statements are also supported by Correia, Flynn, Uliana and Wormald (2007) and Drury (2007).

The adjustments related to GAAP are probably the most controversial and arguable aspect of EVA application. These adjustments are necessary in order to prevent the distortions made by GAAP, especially when applying EVA to companies with divisional structures.

Even though some authors argue that these adjustments are necessary in order to produce earnings figures that are closer to cash flows, other authors often criticize them, for having a small impact on EVA and being very complex and hard to understand (Sharma and Kumar (2010)). As noted by Young (1999), this minor number of adjustments is the result of the skepticism felt by corporate executives, when diverging from GAAP based numbers. Besides this, corporate executives believe that "most of the proposed adjustments have little or no qualitative impact on profits" (Young (1999, p. 9)).

Concerning the non-GAAP adjustments, since EVA is an internal managerial metric, this kind of adjustments often relates to the information that is not publicly available and that impacts mostly the determination of NOPAT—see Weaver (2001) for a survey on the application of these adjustments to the computation of EVA. For example, the divisional NOPAT should be calculated before nominal interest charges. Correia et al. (2007) stated that these adjustments are mainly driven to bring managerial information closer to economic reality.

CHAPTER 3

CHARACTERIZATION OF THE SAMPLE, METHODOLOGY AND EVA ACCOUNTING ADJUSTMENTS

To our best knowledge, this is the first work on the analysis of value creation by the Portuguese SME. Thus, our empirical investigation seeks to answer the central questions "Did the biggest and best Portuguese SME create value during the first three years of the crisis period of 2008 2010?" and "Which sectors have contributed most to the creation of value in this period?".

The methodology adopted for the achievement of this work is a positivist one and consists on the analysis of results obtained after the application of a measurement tool to a sample of Portuguese SME. In section 3.1, we proceed to the characterization of our sample. After that, section 3.2 describes, in full detail, the methodology adopted for the achievement of our work. Finally, section 3.3 describes the necessary adjustments to the financial statements in order to compute EVA.

3.1. Sample Database Description

According to INE, the Portuguese business environment is represented, by more than 99%, of micro, small and medium enterprises. Table 3.1 describes a picture of the Portuguese business environment, at the end of 2007.

The total number of SME in Portugal, in the year of 2007, was superior to 1,100 thousand companies, creating more than 3 million jobs and generating revenues up to $250,000 \text{k} \in$. The micro companies alone, representing 95.5% of the total SME, are responsible for more than 35% of the total revenues. Our sample includes the biggest and best Portuguese SME that had a consistent presence in the TOP 1,000 Ranking, over the 2008-2010 period.

	Number of	%	Number of	%	Total	%
	Companies		Employes		Revenues	
Micro	1,051,195	95.50	$1,\!677,\!446$	54.56	89	35.56
Small	43,443	3.95	$820,\!299$	26.68	85.3	34.08
Medium	6,124	0.55	$576,\!556$	18.75	76	30.36
Total	1,100,762	100.00	3,074,301	100.00	250	100.00

Table 3.1: Portrait of the Portuguese SME in 2007

Table 3.1 reports the number of companies, the number of employees and the total revenues (billion of euros) for the year 2007, segmented by size of SME. Sources: INE, IAPMEI and Diário Económico (2010).

The database, provided by Dun & Bradstreet, contains the financial and accounting statements for those companies, through the years between 2007 and 2010, as selected and published by Revista Exame.

Starting with 4,000 raw records, corresponding to a total number of 1,762 companies, we applied some filtering criteria, with the purpose of obtaining a sufficiently homogeneous and representative sample of the Portuguese business environment. The applied criteria were i) permanency in the ranking through the years of 2007, 2008, 2009, 2010; ii) companies should not provide negative equity value in any year and iii) the ratio of financial costs-to-debt ratio should not exceed 100% in any year.

After applying the mentioned criteria, the sample was reduced to 315 companies, scattered in 9 aggregated (or super) sectors that, in turn, can be decomposed into 18 different sectors. Of these 315 companies, 208 have been classified as "PME Excelência" of the year 2013. Figure 3.1 shows the distribution by sector of the total number of companies in our sample.

The two supersectors "Commerce" and "Manufacturing and Chemical Industries" aggregate a total of 155 companies. "Food Products" and "Food/Staple Retailers" are the most representative sectors, with 48 and 43 companies, respectively. On the other hand, the sector that has the minor representativeness is "Metals and Mining", with only 6 companies. Table 3.2 shows the average yearly key indicators for each sector, with reference to the total period of three years, 2008-2010.



Figure 3.1: Number of companies by sector

Figure 3.1 shows the distribution by sector of the total number of companies in our sample. Supersector "Manufactoring and Chemichal Industries" aggregates the "Paper and Forestry Products", "Auto and Components", "Electrical Equipment", "Machinery, "Pharmaceuticals", "Chemicals" and "Textiles, Apparel and Luxury Goods" sectors and supersector "Commerce" aggregates the "Multiline Retailers", "Specialty Retailers", "Electronic Equipment, Instruments and Components" and "Wholesaling" sectors.

As seen in Figure 3.1, the sectors with more representativeness are "Food Products" and "Food/Staple Retailers", followed by "Multiline Retailers", with 43, 48 and 35 companies present in the ranking for the consecutive three year period, respectively. The sector that registered the lowest number of companies present in the same ranking, for the same period is "Auto and Components", with a number of only 4 companies.

Because "Food Products" is the sector that includes the highest number of companies, it becomes the most representative, independently of the key indicator chosen. In order to understand the average dimension of the companies included in each sector, Table 3.3 shows the average key indicators by company, computed by dividing the key indicator (average for the three year period) by its correspondent number of companies.

Table 3.2: Key indicators by sector (2008-2010)

						'(000 EUR
	No of	Net					Net
	Comp.	Assets	Debt	Equity	Revenues	EBIT	Income
Food Products	48	646,213	259,556	$226,\!052$	1,063,494	27,326	$16,\!150$
Paper and Forestry Products	7	108,781	34,268	$57,\!836$	$133,\!667$	4,011	$2,\!683$
Auto and Components	4	$78,\!971$	$36,\!156$	$26,\!442$	$65,\!859$	2,972	$2,\!331$
Electrical Equipment	6	99,727	$25,\!093$	45,104	$91,\!901$	$8,\!571$	4,832
Machinery	8	175,744	$75,\!561$	57,029	$174,\!468$	$12,\!256$	5,768
Pharmaceuticals	5	61,782	20,997	12,296	$112,\!591$	$2,\!987$	$1,\!921$
Chemicals	10	$164,\!954$	$64,\!687$	64,114	$199,\!249$	$7,\!656$	$5,\!055$
Textiles, Apparel and Luxury Goods	18	$215,\!322$	$77,\!471$	70,982	$326,\!611$	9,740	$4,\!525$
Multiline Retailers	35	$312,\!425$	110,888	89,128	$705,\!202$	$10,\!407$	$11,\!077$
Specialty Retailers	26	$304,\!260$	$110,\!623$	$87,\!435$	$531,\!460$	1,732	6,260
Electronic Eq. Inst. Comp. Retailers	12	124,460	$39,\!974$	38,093	$211,\!697$	$5,\!449$	$3,\!021$
Wholesaling	24	$303,\!479$	$109,\!619$	$102,\!390$	$552,\!573$	$12,\!624$	$6,\!680$
Construction and Engineering	29	$504,\!996$	$194,\!920$	$137,\!188$	$501,\!872$	$16,\!869$	$12,\!107$
Food/Staple Retailers	43	$285,\!212$	89,717	90,713	847,050	$12,\!165$	11,779
Oil and Gas	15	82,519	$21,\!625$	$22,\!612$	$362,\!684$	$3,\!006$	1,932
Metals and Mining	6	$83,\!885$	32,168	32,349	111,977	3,794	$1,\!550$
Commercial Services and Supplies	9	$62,\!499$	18,844	$19,\!658$	$183,\!137$	2,741	$2,\!070$
Transportation	10	114,700	41,779	$36,\!525$	$188,\!686$	$3,\!589$	6,296

Table 3.2 shows the average yearly key indicators (Net Assets, Debt, Equity, Revenues, EBIT and Net income) for each sector, with reference to the total period of three years, 2008-2010. Except for the number of companies, all values are expressed in thousands of euros ('000 EUR).

Actually, in average terms, "Machinery" registered the highest Net Assets and Debt, $22M \in$ and $9.4M \in$ respectively during the three year period of 2008-2010, while the "Oil and Gas" sector registered the lowest values for the same indicators. In terms of Revenues, the average "Oil and Gas" company registered the highest value, of 24.2 M \in , whereas the average "Electrical Equipment" company registered the lowest, $15.3M \in$. When one considers the key indicator Net Income, "Oil and Gas" registered, in average terms, the lowest value, $129k \in$ while "Electrical Equipment" reports the highest, $805k \in$.

Besides analyzing the key indicators for each sector, it is also relevant to understand its evolution on a yearly basis. Table 3.4 reports the descriptive statistics for each year considering the whole sample.

Indicators such as Net Assets and Equity show, however slight, a consecutive yearly increase over the three year period, as well as EBIT and Net Income. On the other hand, Revenues suffered a slight decrease from 2008 to 2009, but on the following period the registered variation

						,C	000 EUR
	No. of	Net					Net
	Comp.	Assets	Debt	Equity	Revenues	EBIT	Income
Food Products	48	13,463	5,407	4,709	$22,\!156$	569	336
Paper and Forestry Products	7	$15,\!540$	$4,\!895$	8,262	$19,\!095$	573	383
Auto and Components	4	19,743	9,039	$6,\!610$	$16,\!465$	743	583
Electrical Equipment	6	$16,\!621$	4,182	7,517	$15,\!317$	$1,\!429$	805
Machinery	8	21,968	$9,\!445$	7,129	$21,\!809$	1,532	721
Pharmaceuticals	5	$12,\!356$	$4,\!199$	$2,\!459$	22,518	597	384
Chemicals	10	$16,\!495$	6,469	6,411	19,925	766	505
Textiles, Apparel and Luxury Goods	18	11,962	$4,\!304$	3,943	$18,\!145$	541	251
Multiline Retailers	35	8,926	3,168	2,547	$20,\!149$	297	316
Specialty Retailers	26	11,702	4,255	3,363	20,441	67	241
Electronic Eq. Inst. Comp. Retailers	12	10,372	$3,\!331$	$3,\!174$	$17,\!641$	454	252
Wholesaling	24	$12,\!645$	4,567	4,266	$23,\!024$	526	278
Construction and Engineering	29	$17,\!414$	6,721	4,731	$17,\!306$	582	417
Food/Staple Retailers	43	$6,\!633$	2,086	2,110	$19,\!699$	283	274
Oil and Gas	15	5,501	$1,\!442$	1,507	$24,\!179$	200	129
Metals and Mining	6	$13,\!981$	5,361	5,392	$18,\!663$	632	258
Commercial Services and Supplies	9	6,944	2,094	2,184	$20,\!349$	305	230
Transportation	10	$11,\!470$	$4,\!178$	$3,\!653$	18,869	359	630
Representative company		11,841	4,330	3,860	20,204	470	337

Table 3.3: Average company key indicators by sector (2008-2010)

....

Table 3.3 shows the average key indicators (Net Assets, Debt, Equity, Revenues, EBIT and Net income) by company, computed by dividing the key indicator (average for the three year period) by the number of companies of each sector. The representative company refers to a theoretical company, which portraits as the average SME of whole sample for each key indicator. Except for the number of companies, all values are expressed in thousands of euros ('000 EUR).

had a positive sign and was greater than the decline recorded in the previous year. The global Debt level decreases from the first to the second year, indicating a slight contraction in the companies' average leverage level; however on the following period, 2009 to 2010, there was a significant increase (about 6 times greater), probably due to a raise in the investment level of net fixed assets.

However, we cannot infer rigorous conclusions from these statistics, given that 2010 is a very early year of the financial crisis and its effects were still only beginning to take a toll on the Portuguese society.

After filtering the data, we proceeded to the organization of the accounting statements in order to obtain the three main components for the calculation of EVA. The results are presented in chapter 4.

						'000 EUR
	Net					Net
	Assets	Debt	Equity	Revenues	EBIT	Income
Panel A: 2008						
Average	$11,\!347$	4,113	3,529	20,516	452	318
Std Deviation	7,381	3,971	$3,\!067$	7,252	964	510
1st Quartile	5,779	1,281	1,366	15,141	56	53
2nd Quartile	10,326	2,868	2,709	18,280	338	166
3rd Quartile	15,318	5,929	$4,\!594$	24,496	733	441
Maximum	39,043	23,910	19,920	46,440	$5,\!673$	3,272
Minimum	706	5	178	11,757	-3,583	-1,072
Panel B: 2009						
Average	$11,\!590$	3,881	$3,\!835$	19,319	475	334
Std Deviation	7,452	3,903	3,232	7,275	922	580
1st Quartile	5,751	$1,\!153$	$1,\!489$	13,734	72	56
2nd Quartile	$10,\!157$	$2,\!637$	2,947	17,269	316	175
3rd Quartile	$15,\!640$	5,337	5,215	23,137	785	448
Maximum	$38,\!873$	21,249	21,265	44,217	6,752	$3,\!287$
Minimum	791	0	182	10,146	-3,924	-1,710
Panel C: 2010						
Average	$12,\!586$	4,996	4,216	20,776	482	358
Std Deviation	8,188	4,667	3,462	8,343	858	527
1st Quartile	6,367	$1,\!651$	1,733	14,771	79	74
2nd Quartile	11,018	3,377	$3,\!241$	18,372	327	210
3rd Quartile	17,056	6,814	5,527	25,065	892	539
Maximum	41,463	22,591	$21,\!622$	48,794	$5,\!438$	3,254
Minimum	964	0	153	10,260	-3,679	-1,546
Representative company	11,841	4,330	3,860	20,204	470	337

Table 3.4: Descriptive statistics for the key indicators (2008, 2009 and 2010)

Table 3.4 reports the descriptive statistics for the key indicators (Net Assets, Debt, Equity, Revenues, EBIT and Net income (thousands of euros)), for each year, considering the whole sample. The representative company refers to a theoretical company, which portraits as the average SME of the whole sample for each key indicator.

3.2. Methodology

In the present section we illustrate, in full detail, the methodology applied to compute EVA from the data previously described. We have divided the computation of EVA in three main components: i) invested capital, at the beginning of each year, ii) cost of capital, which is the weighted average cost of capital and iii) return on invested capital.

3.2.1. Invested Capital

The invested capital (IC) is the amount of money that holders and shareholders have committed to the company. Basically, and given the concept of EVA, it is the amount of money invested in the company for which management has to produce a return.

IC (using an operating approach) can be estimated by subtracting all non-interest-bearing current liabilities (net working capital) from total liabilities and total equity (or total assets). In other words, IC is the sum of fixed (tangible and intangible) assets, plus investments, cash and working capital requirements. This was our first approach to the estimation of IC.

However, due to the lack of relevant data and detailed information to properly compute working capital, our estimations were distorted. Thus, we decided to follow an alternative methodology as suggested by Stewart (1991) and reinforced by Roztocki and Needy (1999). According to these authors, a simpler way to estimate a company's IC is to sum all of its financial sources, such as short-term debt and long-term debt, and owners' equity.

This methodology is considered the most simple, since we only need the liabilities' side of the balance sheet¹. Hence, IC in each firm is the sum of shareholders' equity plus total liabilities, net of non-interest-bearing short-term liabilities, i.e. amongst the total current liabilities we have only considered short-term financial debt.

Even though the companies present in our sample were not affected by such conditions, one must be attentive of the long term non-interest-bearing liabilities. For those calculating EVA, provisions, deferred taxes and minority interests should be considered equity equivalents and taken into account for the necessary adjustments.

3.2.2. Cost of Capital

Our next step was to compute the average cost of capital. Typically, companies resort to two sources of financing: using shareholders' and investors' capital, which translates into equity,

¹All the money on the liabilities side of the balance sheet is money committed to the company, although part of it was not actually invested in the company with the purpose of obtaining a return. Included in this later case, are accounts payable to suppliers, money owed to employees or to the state and other short term liabilities, with the exception of short-term debt.

Operating View Point	Financing View Point
Cash	
Working	Common Equity
Capital	Preferred Equity
	Minority Interessts
	Equity Equivalents
Fixed	
Assets	Long Term
	Interest Bearing
	Liabilities
Investments	Short Term Liabilities
Assets side	Liabilities side

Figure 3.2: Invested Capital

Figure 3.2 shows a comparison between the two approaches to calculating IC: i) operating view point, using the right side (Assets) or ii) financing view point, using the left side (Liabilities) of the balance sheet.

and borrowing capital from financial institutions (such as banks), which translates into financial debt. Both types of financing have an associated cost, to which we call cost of equity and cost of debt, respectively. The total cost of a company's financing can be estimated by the weighted average of these two factors (WACC), using the formula:

$$WACC = r_e \times \frac{E}{IC} + r_d \times \frac{D}{IC} \times (1 - t_c), \qquad (3.1)$$

where r_e represents the cost of equity, r_d is the cost of debt, E is the book value of the company's equity, D is the book value of the company's financial debt, IC is the invested capital, as a result of (D + E), and t_c represents the corporate tax rate, which was obtained via Ministry of Finance's website. It should be noted that, if the companies in our sample were listed companies, we should use the market values for debt and equity, as the WACC methodology postulates.

Capital structure

In what concerns the financial structure of the companies, it is necessary to compute the market value of the firm, and the breakdown between equity and debt. The breakdown between equity and debt should be computed on a market value basis rather than in a book value basis. This is because it is only the market value that truly reflects the cost of funding².

However, as referred above, since the companies present in our sample are not listed companies, we have to use the companies' capital structure weightings on a book value basis, and assume it as a proxy value for its possible market value. In a survey by Weaver (2001), the author asked 30 companies whether they look at the capital structure weightings on a book or market basis. Fourteen companies used market weights, compared to eleven companies that used book weights (the remaining five companies did not answer). It should be noted that allowances for guarantee, doubtful accounts, contingency values, deferred taxes and (possible) minority interests were implicitly included in the equity book value.

Cost of debt

The average cost of debt may be estimated using the yield to maturity of the company's bond issues or through the use of debt rating tables. Likewise the capital structure weightings, the cost of debt should be the market cost of debt, i.e. the cost that the company would have to pay so as to issue new debt³.

Because we do not have market values for the debt of the companies in our sample, the cost of debt was estimated through the ratio of interest paid and the amount of financial debt for each year and, therefore, we considered the cost of debt as an average interest rate.

Cost of equity

The cost of equity is very difficult to observe as it is the return demanded by the market for a stock with a risk similar to the company being evaluated. It is common practice to use the

²For instance, if a company wants to change its financial structure to, suppose, increase the weight of debt and reduce the weight of equity, it will have to issue bonds at market prices and/or repurchase stocks at market prices. Thus, weighting the cost of capital with market values is the only way to obtain the true cost of capital. ³This is the actual opportunity cost for bond holders, and thus is what they are expecting to obtain by investing their money in similar risk projects.

Capital Asset Pricing Model (CAPM) to estimate the cost of equity. In fact, Weaver (2001) reported in his survey that 29 of the 30 companies employ CAPM to estimate their cost of equity.

The CAPM model postulates that the cost of equity is computed as the sum of the risk free rate, plus a risk premium that depends on the market risk premium and a beta coefficient that reflects the systematic risk of the company.

$$r_e = r_f + \beta_L \times (r_m - r_f), \qquad (3.2)$$

where r_f is the risk free rate, β_L is the levered beta and $(r_m - r_f)$ represents the market risk premium.

Risk Free Rates. As the risk free rate is observable in the yield to maturity of long term government bonds, for each year we used the one year yield benchmark for the Eurozone, obtained via Bloomberg. Now, we are left with the problem of estimating the market risk premium and the beta of the company.

Market Risk Premium. The market risk premium, at first calculated by the difference between the Eurozone stockmarket benchmark (STOXX600) and the the riskless rates for the Eurozone (treasury bills with one year maturity), as mentioned in the above paragraph. However, given the high volatility of the values that we have obtained, highly influenced by the uncertainty of the economical and financial environment experienced all across Europe, we have decided to use an alternative approach, resorting to the work of Professor Pablo Fernandez, from IESE Business School (Fernandez, Aguirreamalloa and Linares (2013), Fernandez and Del Campo (2010)).

Professor Fernandez conducted several studies on equity risk premium, based on surveys sent to different professionals of the economic and financial areas (professors, risk analysts, managers of companies and managers of financial companies) aiming to produce a sensible estimate for the market risk premium of several European countries, including Portugal.

As displayed in Table 3.5, the effectively used values for the market risk premium in the years of 2008, 2009 and 2010 are 5.3%, 5.3% and 5.15%, respectively. The market risk premium

	2008	2009	2010
Author's estimates (STOXX600)	-52.70%	23.42%	6.34%
Fernandez et al. (2013)	5.30%	5.30%	5.3% and $5%$
Values Effectively Used	5.30%	5.30%	5.15%

Table 3.5: Estimates for the market risk premium

Table 3.5 shows the estimates for the market risk premium collected from different sources. The first line shows the author's estimates based on the difference between STOXX600 benchmark and the risk free rate; the second line shows the estimates obtained from Fernandez et al. (2013). Both lines refer to European market risk premiums. The third line reports the effectively used estimates for the market risk premium, in the years 2008, 2009 and 2010; the 2010 estimate results from the average between 5.3% and 5.15%.

for the year 2010 is an average of the results found by Fernandez et al. (2013) in the surveys made to both professors (5.3%) and risk analysts (5%) for the European risk premium. Professor Fernandez has also estimated the market risk premiums for Portugal. However, since we are using the European MSCI Index as a benchmark, for a matter of consistency, the estimates for Europe were the most appropriate values to use. The adjustments for the Portuguese context will be made through the levered betas.

Company's systematic risk (beta). The best way to obtain an estimate of the companies' levered beta is to use past stock return information and regress it versus the local market. For companies not listed, the best way to calculate a beta is to start with a beta of a company operating in the same line of business and adjusting it for possible differences in capital structure. This process was, amongst all the components needed for the cost of capital, the most complex and, for simplicity's sake, can be decomposed in several steps. Since we did not have comparable companies, we had to use the sectorial information to proxy the average risk of a company operating in that sector. Due to the absence of historical information with sufficient depth on the betas for each sector of activity in Portugal (even in the cases where there was some historical depth, it only went as far as a year or two in time), we had to estimate them.

Step 1: We started by collecting the monthly values for the MSCI Index for the European activity sectors. This step consisted in regressing the sectorial returns versus the market portfolio returns via the market model⁴ to obtain the sectorial levered betas, assuming as the

 $^{^{4}}$ The market model used was

market portfolio the MSCI Europe Index. We chose the MSCI Europe Index as a market portfolio mainly for three reasons: first, it is calculated based on the weighted average of the stock prices; second, it is a benchmark that incorporates income generated from the distribution of dividends and, finally, it is a geographically diversified index, as it incorporates in its composition shares from companies scattered from all across Europe.

Step 2: After computing the levered beta for each sector, two procedures are necessary for the betas to reflect different capital structures.

First, we had to unlever the betas, which means we had to calculate the beta of the company as if it had no debt. In order to do so, we used annual averages of debt-to-equity ratios for each sector⁵, obtained via Bloomberg. For this same purpose, we used income corporate taxes in effect, collected from Eurostat⁶. Table 3.6 displays the estimates of the levered betas, debt-toequity ratios and the implicit unlevered betas of the European sectors for years of 2008, 2009 and 2010. Without getting into much technical detail, the unlever and re-lever of the beta can be done through the equation 3.3^7 . For more details, please see Hamada (1972).

$$\beta_U = \frac{\beta_L}{1 + (1 - t_c) \times \frac{D}{E}},\tag{3.3}$$

where β_U and β_L are the unlevered and levered betas, respectively, E is the book value of equity and D is the book value of debt (assumed here as proxies for their respective market values); t_c is the corporate tax rate.

Since the estimated unlevered betas reflect the average risk for the European sectors, which are much lower than those in Portugal, it was necessary to make another adjustment, through the estimation of a country risk coefficient for Portugal⁸.

where r_{m_i} , r_{i_i} are the logatithmic rates of return for the market m and the sector i, respectively, α_i is the the component of the return that is independent of the market's behavior, β_i is the levered beta for the sector i and ε_{i_t} is the residual estimation error, which we assume that has an expected value of zero and homoscedastic variance.

⁵The annual debt-to-equity ratios used were computed based on the trimestral debt-to-equity ratios for each year, obtained directly via Bloomberg.

⁶The corporate tax rates were retrieved from an Eurostat report Eurostat (2011).

⁷The beta that reflects the debts' risk for the sector was considered zero, simulating the most neutral situation possible.

⁸The purpose of this risk coefficient aims to proxy the risk of Portugal, compared to the risk of other Eurozone countries. It was computed based on the 5 year maturity yields for these same countries.

Corporate Tax (%)	23.60				23.50			23.30		
	2008			2009			2010			
	β_L	D/E	β_U	β_L	D/E	β_U	β_L	D/E	β_U	
Food Products	0.55	72.21	0.35	0.44	70.54	0.29	0.43	65.20	0.29	
Paper and Forestry Products	1.08	55.59	0.76	1.20	58.76	0.83	1.32	58.20	0.91	
Auto and Components	1.35	57.51	0.94	1.18	51.59	0.84	1.15	52.88	0.82	
Electrical Equipment	0.76	50.96	0.55	1.11	53.16	0.79	0.76	52.99	0.54	
Machinery	1.65	98.16	0.94	1.61	100.32	0.91	1.51	99.01	0.86	
Pharmaceuticals	0.34	56.54	0.23	0.30	54.79	0.21	0.36	56.74	0.25	
Chemicals	0.93	51.49	0.66	1.02	51.89	0.73	1.01	52.40	0.72	
Textiles, Apparel and Luxury Goods	1.31	21.56	1.12	1.24	20.54	1.07	1.25	21.52	1.07	
Multiline Retailers	1.33	76.19	0.84	1.06	68.80	0.69	1.10	77.91	0.69	
Specialty Retailers	0.98	7.72	0.93	0.75	7.33	0.71	0.73	7.64	0.69	
Electronic Eq. Inst. Comp. Retailers	0.54	78.64	0.34	0.54	73.07	0.34	0.54	76.55	0.34	
Wholesaling	0.62	100.97	0.35	0.58	99.27	0.33	0.52	96.69	0.30	
Construction and Engineering	1.34	158.73	0.60	1.24	145.39	0.59	1.23	154.07	0.56	
Food/Staple Retailers	0.78	80.86	0.48	0.71	84.92	0.43	0.64	89.75	0.38	
Oil and Gas	1.04	39.67	0.80	0.76	38.53	0.59	0.76	39.06	0.59	
Metals and Mining	1.35	35.21	1.06	1.63	39.25	1.26	1.63	37.79	1.27	
Commercial Services and Supplies	0.84	124.06	0.43	0.70	127.45	0.36	0.68	126.22	0.35	
Transportation	0.96	386.85	0.24	1.18	303.82	0.35	1.19	217.06	0.45	

 Table 3.6: Levered and unlevered beta for the European sectors

Table 3.6 displays the estimates of the levered betas (β_L) , debt-to-equity ratios $(\frac{D}{E})$ and the implicit unlevered betas (β_U) of the European sectors for years of 2008, 2009 and 2010, computed using equation 3.3 and assuming that beta of debt (β_D) is zero.

Godfrey and Espinosa (1996) suggested a method in which we could add the country default spread to the risk free rate and multiply the European risk premium by the volatility of the company's equity market relative to the European market. Due to the significant correlation between the European and the Portuguese markets, this method did not seem to be the best way to introduce the country market risk premium. Thus, we tried to find alternative methods to incorporate the country market risk premium in the cost of equity. We chose to estimate the country risk premium as a spread over the benchmark treasury yield.

For each country, the sovereign risk premium was computed as the spread between its sovereign five year yield and the Eurozone benchmark yield for the same maturity. After estimating the sovereign risk premium for each country, we were able to calculate an average of the sovereign risk premium, which we will consider as the European risk premium. Finally, the country risk coefficient for Portugal is the ratio between the Portuguese risk premium and the average Eurozone sovereign risk premium. This risk coefficient, as displayed in Table 3.7, will be used to adjust the European unlevered betas into the Portuguese unlevered betas for each sector.

	2007	2008	2009	2010
(1) Portuguese risk premium (b.p.)	8	58	220	381
(2) European risk premium (b.p.)	10	46	215	282
(3) = (1) / (2) Coefficient risk premium	1.23	1.24	1.02	1.35

Table 3.7: Estimates for country risk coefficient

Table 3.7 shows the risk coefficient that will be used to estimate the unlevered betas (β_U) for Portuguese sectors. The Portuguese risk premium is the difference between the Portuguese five year yield and the five year European benchmark; Europe risk premium is the difference between the average of the five year yields of the European countries and the European benchmark. The European benchmak considered is the Bloomerg's European sovereign five year yield index.

As observed in Table 3.7, Portugal has followed the increase tendency of Eurozone's spread over the five year yield sovereign benchmark, increasing from 10 b.p. in 2007 to 282 b.p. in 2010. Regarding the Eurozone countries, the five year yield average spread rose from 8 b.p. in 2007 to 381 b.p. in 2010. This evolution is particularly influenced by the increase in credit spreads, not only in Portugal, but also in countries such as Greece and Ireland, due to the economic and financial bailout in 2010. The Portuguese coefficient risk premium, which was relatively stable during the years of 2007 and 2008, has decreased to 1.02 in 2009 and registered an increase to 1.35 in 2010.

Secondly, we had to re-lever the beta, i.e. we had to recalculate the levered betas, starting from the unlevered beta that reflects the assets' average risk for a company and adjusting for the capital structure of each company, in each sector. This process was guided by Damodaran (2001) bottom-up betas approach.

3.2.3. Return on Invested Capital

As WACC reflects the average capital charges, the return on invested capital (ROIC) reflects the return on capital employed in the business. ROIC was computed based on equation 3.4:

$$ROIC = \frac{EBIT \left(1 - t_c\right)}{IC_{boy}},\tag{3.4}$$

	2008		20	09	2010		
	β_U Eur	$\beta_U \operatorname{Prt}$	β_U Eur	$\beta_U \operatorname{Prt}$	β_U Eur	$\beta_U \operatorname{Prt}$	
Food Products	0.35	0.44	0.29	0.29	0.29	0.39	
Paper and Forestry Products	0.76	0.94	0.83	0.85	0.91	1.23	
Auto and Components	0.94	1.17	0.84	0.86	0.82	1.11	
Electrical Equipment	0.55	0.68	0.79	0.81	0.54	0.73	
Machinery	0.94	1.17	0.91	0.94	0.86	1.16	
Pharmaceuticals	0.23	0.29	0.21	0.22	0.25	0.34	
Chemicals	0.66	0.83	0.73	0.75	0.72	0.97	
Textiles, Apparel and Luxury Goods	1.12	1.40	1.07	1.10	1.07	1.44	
Multiline Retailers	0.84	1.05	0.69	0.71	0.69	0.93	
Specialty Retailers	0.93	1.15	0.71	0.73	0.69	0.93	
Electronic Eq. Inst. Comp. Retailers	0.34	0.42	0.34	0.35	0.34	0.46	
Wholesaling	0.35	0.43	0.33	0.34	0.30	0.40	
Construction and Engineering	0.60	0.75	0.59	0.60	0.56	0.76	
Food/Staple Retailers	0.48	0.60	0.43	0.44	0.38	0.51	
Oil and Gas	0.80	0.99	0.59	0.60	0.59	0.79	
Metals and Mining	1.06	1.32	1.26	1.29	1.27	1.71	
Commercial Services and Supplies	0.43	0.54	0.36	0.36	0.35	0.47	
Transportation	0.24	0.30	0.35	0.36	0.45	0.61	

Table 3.8: Estimated unlevered betas: Europe and Portugal

Table 3.8 exhibits the unlevered betas estimated for the European sectors and Portuguese sectors, which were adjusted by the Portuguese country risk coefficient, shown in Table 3.7. " β_U Eur" refers to the unlevered betas of the European sectors and " β_U Prt" refers to the unlevered betas of the Portuguese sectors.

where EBIT corresponds to the earnings before interest and taxes, t_c is the corporate tax in effect for the period and IC_{boy} is the invested capital at the beginning of the year.

3.2.4. Economic Value Added

Having calculated the three main components needed in order to estimate EVA, we have now gathered all the information we need so as to proceed with our work. To do so, we have applied equation 3.5.

$$EVA = (ROIC - WACC) \times IC_{boy}.$$
(3.5)

Equation 3.5 was applied to all the companies, providing an estimate for each company's EVA, in absolute terms. Because the companies present in our rankings belong to very different sectors, with a very different set of financial and economic characteristics, these values were hardly comparable and might lead us into taking the wrong conclusions. In order to have

comparable data between companies from different sectors and different time spans, we decided to use a different indicator, to which we called EVA spread that, because of its relative nature, allowed us to make the comparisons needed between companies, in order to estimate the best performances in each year and between each sector of activity.

$$EVA \ Spread = ROIC - WACC. \tag{3.6}$$

As shown in the equation 3.6, EVA Spread is easy to compute as it is given by the difference between the return on invested capital and the weighted average cost of capital.

3.2.5. Market Value Added

According to Dierks and Patel (1997), the Market Value Added (MVA) is an indicator, in absolute terms, of a company's created wealth for its investors over a period of time. The computation of the MVA is as simple as the sum of the EVAs registered over a certain period of time, as in equation 3.7:

$$MVA = \sum_{t=1}^{n} EVA_t, \tag{3.7}$$

where MVA is the Market Value Added and EVA_t is the estimated Economic Value Added for the year t. So, in our case, MVA is the sum of EVAs for 2008-2010 period.

As in the previous indicator, EVA, there was a need to adapt MVA, which gave us absolute values that might distort our conclusions. Thus, we decided to transform the absolute MVA into a relative basis, as shown by equation 3.8.

$$RMVA = \frac{MVA}{\frac{1}{n}\sum_{t=1}^{n}IC_t},$$
(3.8)

The Relative MVA (RMVA) is estimated by dividing a company's MVA for a period of time t, by the average of its IC during the same period (in our sample 2008-2010, or better

 $2007\ 2009$ since IC is lagged – i.e., it refers to the beginning of the year, which is the same as the end of the previous year).

3.3. EVA accounting adjustments

The adjustments recommended by authors such as Stewart (1991), Young (1999), Weaver (2001), Correia et al. (2007) and Drury (2007) should be made only if the four following criteria, as mentioned in section 2.4, are met: i) the amounts considered are significant; ii) they are likely to have a material influence; iii) operational level employees can easily understand them; and iv) the required information is easily tracked.

These criteria are utmost important when one is dealing with companies with a divisional structure. The more complex the structure, the greater the possible distortions made by the application of GAAP. In order to apply these adjustments, one must have access to the internal accounting, financial and other managerial information. Since the present dissertation aims to measure the value created by the Portuguese SME, using a representative sample of 315 companies, evaluating the adequacy of these criteria to each company would be virtually impossible.

The computation of EVA, according to equation 3.5, which seems straightforward and simple, was applied to all the 315 companies in the sample. However, we decided to take on some assumptions that aimed to approximate the available information to the economic reality. Such assumptions were i) EBIT after taxes as a proxy for NOPAT; ii) IC—as a result of the sum of the book value of the company's equity and the book value of the company's financial debt, only reflects the value of the assets in place; iii) the discretionary expenses such as research and development and market, and advertising are implicitly included in the IC; iv) the owners' loans were considered as financial debt; v) financial leasings were also considered as part of the financial debt, by the inclusion of the accounting item fixed asset suppliers; and vi) non-recurrent gains and losses were not considered⁹.

⁹According to Frezati (2003), single events whose repercussion on income only occurs at a specific moment in time, should not be included in the computation of income. Van der Poll et al. (2011) also noted that, in what concerns NOPAT, non-recurrent gains and losses should not be included in the EVA computation. In our sample, the non-recurrent gains and losses represent about 20% of the taxable income.

Adjustments that relate to i) the goodwill and respective amortization; ii) deferred taxes iii) depreciation methods; iv) inventory valuation methods; v) allowance for guarantee, doubtful accounts and contingency are also relevant, however they were not considered in the computation of EVA due to the lack of information.

Chapter 4 presents the empirical results of the EVA calculation for the sample of 315 biggest and best Portuguese SME.

CHAPTER 4

PERFORMANCE ASSESSMENT RESULTS

In this chapter, we report the results of our performance assessment, as measured by EVA and MVA, of the 315 SME analyzed in our sample for the period 2008 2010. Table 4.1 summarizes these results.

Panel A: EVA	2008	2009	2010
Number of companies with negative EVA	216	197	157
Sum of negative EVAs	$-119,\!646$	-96,795	-79,118
Number of companies with positive EVA	99	118	158
Sum of positive EVAs	39,878	46,971	$70,\!681$
Total EVA	-79,769	-49,824	-8,437
Panel B: MVA	2	008 - 2010	
Number of companies with negative MVA		200	
Sum of negative MVAs		-262,968	
Number of companies with positive MVA		115	
Sum of positive MVAs		$124,\!938$	
Total MVA		-138,029	

Table 4.1: Summary of EVA and MVA results

Table 4.1 exhibits the results of the performance assessment, as measured by EVA and MVA, of the 315 SMEs analyzed in our sample for the period 2008 2010.

Globally, the total EVA is negative in each of the three years. The total sum of negative EVAs is $-119,646k \in$ in 2008, $96,795k \in$ in 2009 and $-79,118k \in$ in 2010. However, the number of companies with negative EVA decreases by each year, from 216 companies in 2008, to 197 companies in 2009, to 157 companies in 2010. In line with these results, the number of companies with negative MVA is 200, amounting to $-262,968k \in$. The total MVA is negative and adds up to $-138,029k \in$.

In section 4.1 we analyze the three main EVA components and the results of the EVA evaluation, as presented in subsection 3.2.4, equation 3.5. In section 4.2, we try to answer our investigation question "Did the biggest and best Portuguese SME create value during the first three years of the crisis period of 2008 2010?" by examining MVA and RMVA estimated for

the whole sample, and detailed by sector. Finally, in section 4.3 we rank the ten companies with the best and worst performances, measured in terms of EVA Spread and RMVA.

4.1. EVA components and EVA Spread analysis

Table 4.2 displays the descriptive statistics for the three main EVA components, as well as our estimates of EVA and EVA Spread. Panel A, Panel B and Panel C present the global figures for the years of 2008, 2009 and 2010, respectively.

Table 4.2: Descriptive statistics for EVA main components (2008, 2009 and 2010)

					EVA
	WACC	ROIC	IC	EVA	Spread
Panel A: 2008					
Average	9.23%	6.39%	$6,\!584$	-253	-2.84%
Std Deviation	2.96%	14.25%	$5,\!129$	732	14.50%
1st Quartile	7.22%	1.22%	$2,\!648$	-499	-8.14%
2nd Quartile	8.84%	5.14%	5,310	-151	-3.26%
3rd Quartile	10.59%	11.21%	8,840	74	1.60%
Maximum	24.73%	62.29%	$28,\!807$	3,284	56.20%
Minimum	4.46%	-78.85%	386	-4,347	-103.58%
Panel B: 2009					
Average	6.68%	5.05%	$7,\!642$	-158	-1.13%
Std Deviation	2.84%	11.31%	5,925	685	11.62%
1st Quartile	5.01%	1.28%	3,132	-421	-5.84%
2nd Quartile	6.36%	4.53%	6,131	-78	-2.07%
3rd Quartile	7.57%	8.79%	10,246	114	2.42%
Maximum	34.28%	51.43%	34,464	2,249	45.97%
Minimum	2.63%	-88.74%	351	-3,311	-99.84%
Panel C: 2010					
Average	5.10%	6.21%	7,716	-27	2.67%
Std Deviation	1.80%	11.43%	$5,\!892$	734	11.89%
1st Quartile	3.44%	1.62%	3,038	-324	-4.45%
2nd Quartile	4.88%	4.98%	$6,\!341$	0	0.00%
3rd Quartile	6.29%	9.21%	10,108	242	4.80%
Maximum	11.88%	71.48%	$33,\!524$	3,333	69.68%
Minimum	1.80%	-57.21%	421	-4,375	-60.26%
Representative	7.00%	5.88%	7,314	-146	-0.43%

Table 4.2 displays the descriptive statistics for the three main EVA components, as well as our estimates of EVA and EVA Spread. Panel A, Panel B and Panel C present the global figures for the years of 2008, 2009 and 2010, respectively. The representative company refers to a theoretical company, which portraits as the average SME of our whole sample. The EVA components for the representative company result of the sum of each year's average EVA components.

When comparing the results of the three panels, there was a straight decrease in the average WACC from 9.23% in 2008 to 5.10% in 2010. Considering the three years, the maximum observed value was 34.28% in 2009 and the minimum was 1.80% in 2010.

Regarding ROIC, there was no specific tendency noted for the evolution of this variable. In fact, and in average terms, there was a reduction in 2009 and an increase in 2010. The observed evolution may have its justification in the fact that, from 2008 to 2009, both EBIT and IC showed, however positive, a disproportionate variation—that is, IC grew about 3 times more than EBIT. By contrast, between 2009 and 2010, EBIT's growth was more accentuated than that of the IC, reflecting an increase in ROIC over the referred period. The evolution of IC was, for the three years, relatively homogeneous for the three quartiles.

The average EVA estimated for the companies in our sample was negative, in all three years. However, as presented in Table 4.2 this variable displays a continuous and significant improvement during the three year period. In fact, the average EVA was $-253k \in$ in 2008, $-158k \in$ in 2009 and $-27k \in$ in 2010. The maximum and minimum EVA values were observed in 2010, $3,333k \in$ and $-4,375k \in$, respectively. In 2008 and 2009, more than half the companies of our sample reported a negative EVA. In 2010, 50% of the companies reported a negative EVA. The difference, in terms of EVA, between the best company in the first quartile and the worst in the third quartile was, on average, approximately $550k \in$.

Table 4.3 presents a deeper insight on the companies' weighted average EVA components in a sectorial perspective for the years of 2008, 2009 and 2010. All the components are weighted by each sector's IC.

Since EVA Spread reflects the difference between ROIC and WACC, the behavior of this variable is a direct consequence of the combination of the last two. The difference registered between the average EVA and average EVA Spread lies in the different levels of invested capital by each sector. This means that the true economic profit, as described by Dierks and Patel (1997), reflects how well each sector can capitalize their investments, after deducting all the costs inherent to the business. Thus, the analysis of Table 4.3 is going to be made, essentially, in an EVA Spread perspective.

	WACC			ROIC			IC ('000 EUR)			EVA Spread		
	2008	2009	2010	2008	2009	2010	2008	2009	2010	2008	2009	2010
Food Products	3.39%	3.68%	4.61%	7.99%	5.09%	3.29%	8,361	$9,\!451$	9,742	-4.60%	-1.41%	1.32%
Paper and Forestry Products	0.63%	0.69%	0.49%	1.51%	1.10%	0.94%	$12,\!123$	$12,\!358$	12,702	-0.88%	-0.41%	-0.44%
Auto and Components	1.21%	0.41%	0.25%	0.92%	0.72%	0.50%	$12,\!436$	14,711	$14,\!632$	0.29%	-0.31%	-0.25%
Electrical Equipment	1.80%	1.53%	1.01%	1.16%	0.94%	0.60%	9,785	$10,\!974$	$11,\!317$	0.64%	0.59%	0.40%
Machinery	2.27%	2.03%	0.75%	2.12%	1.83%	1.08%	12,068	$15,\!514$	$15,\!693$	0.15%	0.20%	-0.33%
Pharmaceuticals	1.04%	0.08%	1.94%	1.18%	1.02%	0.26%	$5,\!963$	$6,\!901$	$4,\!654$	-0.14%	-0.94%	1.69%
Chemicals	1.81%	2.02%	0.73%	2.32%	1.78%	1.18%	11,019	$12,\!244$	12,039	-0.51%	0.24%	-0.45%
Textiles, Apparel and Luxury Goods	2.91%	2.87%	1.28%	5.80%	4.11%	2.76%	$5,\!936$	$7,\!103$	$7,\!481$	-2.89%	-1.24%	-1.47%
Multiline Retailers	4.45%	4.67%	3.75%	8.91%	5.91%	3.95%	$4,\!157$	$5,\!350$	$5,\!654$	-4.46%	-1.25%	-0.20%
Specialty Retailers	0.06%	0.40%	1.05%	6.62%	4.66%	3.21%	6,519	$7,\!381$	$7,\!271$	-6.56%	-4.25%	-2.15%
Electronic Eq. Inst. Comp Retailers	3.11%	2.72%	2.20%	2.27%	1.57%	0.98%	$5,\!617$	$6,\!349$	6,239	0.83%	1.15%	1.22%
Wholesaling	3.45%	2.33%	3.36%	4.50%	2.93%	1.68%	7,781	$8,\!389$	$8,\!571$	-1.05%	-0.59%	1.69%
Construction and Engineering	3.70%	3.29%	3.68%	5.94%	4.48%	3.15%	8,208	10,283	$10,\!456$	-2.24%	-1.19%	0.53%
Food/Staple Retailers	8.06%	4.21%	6.70%	9.66%	7.38%	3.35%	$3,\!377$	$3,\!966$	$3,\!893$	-1.59%	-3.17%	3.35%
Oil and Gas	2.93%	2.74%	2.37%	3.71%	2.72%	1.75%	2,446	2,750	$2,\!693$	-0.78%	0.03%	0.62%
Metals and Mining	1.50%	0.93%	0.90%	1.65%	1.35%	1.09%	$8,\!673$	9,716	9,777	-0.14%	-0.42%	-0.19%
Commercial Services and Supplies	2.23%	0.97%	1.08%	1.58%	1.10%	0.65%	$3,\!447$	4,214	$3,\!987$	0.65%	-0.13%	0.43%
Transportation	3.51%	2.09%	1.80%	1.60%	1.19%	0.75%	6,383	7,065	7,066	1.91%	0.89%	1.04%

Table 4.3: EVA main components by sector (2008, 2009 and 2010)

Table 4.3 displays the companies' weighted average EVA components in a sectorial perspective for the years of 2008, 2009 and 2010. All the components are weighted by each sector's IC. Since EVA Spread reflects the difference between ROIC and WACC, the behavior of this variable is a direct consequence of the combination of the last two. The difference registered between the average EVA and average EVA Spread lies in the different levels of invested capital by each sector.

Three sectors stand out positively, i.e., present positive EVA Spreads for the three years— "Transportation", "Electronic Equipment, Instruments and Components" and "Electrical Equipment". These three sectors have obtained relatively lower funding costs, while achieving higher operational margins. As opposed, four sectors stand out with negative EVA Spreads for the three years—"Specialty Retailers", "Multiline Retailers", "Textiles, Apparel and Luxury Goods", "Paper and Forestry Products" and "Metals and Mining". The remaining 10 sectors oscillate between positive and negative EVA Spreads along the three year period.

Sectors such as "Pharmaceuticals", "Wholesaling", "Food Products" and "Construction and Engineering" are very sensitive to fluctuations in private consumption. Hence, the improvement registered, in terms of EVA Spread, by the year 2010, might be explained by the increase in private consumption. In fact, this increase might have led to an increase of the operational margins and, henceforth, ROIC. "Pharmaceuticals" sector exhibits some particular characteristics that may have influenced the registered evolution, mainly influenced by the strong decline of WACC (from 1.02% to 0.26%) and a substantial raise of ROIC (from 0.08% to 1.94%) in 2010.

This particular behavior could be explained by the fact that this sector has always been subject to special market conditions, due to the fact that it is a strongly regulated sector with great demographic exposure. Regarding ROIC's evolution, the aftershocks of the economic crisis, along with the entry of the generic pharmaceutical products, forced companies of this sector to practice lower prices and, consequentially, much lower margins than in the previous years. Concerning WACC's behavior, given the funding difficulties, the "Pharmaceuticals" companies were forced to contract loans with their shareholders, mostly without explicit interest, leading to an increase of the debt (without an associated increase in terms of financial costs) and, hence, influencing the weights in the WACC computation. Figure 4.1 reflects the annual evolution of the global weighted average EVA Spread. Even though EVA Spread is negative for the three years, it exhibits a continuous increase in its value. This evolution is explained by both an increase of ROIC and a decrease of WACC.



Figure 4.1: Weighted average EVA Spread, ROIC and WACC (2008, 2009 and 2010)

---EVA Spread ---ROIC ---WACC

Figure 4.1 displays the annual evolution of the global weighted average EVA Spread, ROIC and WACC, using the IC as the weighting factor.

4.2. MVA and RMVA analysis

The MVA, as described by equation 3.7, allows us to compute a company's cumulated economic value creation (or destruction) along several periods of activity. Since different sectors have different levels of IC, their performances cannot be compared in a MVA basis. In order to overcome this difficulty, we used RMVA, as presented in equation 3.8, to purge the effect of the IC on the MVA computation. Table 4.4 exhibits the global MVA of all companies, and respective RMVA, descriptive statistics.

During the period 2008-2010, as we have seen in Table 4.2, the companies, as a whole, reported a global average MVA of $-438k \in$ or, in a RMVA basis, -3.70%. The first quartile of our sample reported MVAs lower than $-1,111k \in$, while the worst company of the third quartile reports an MVA higher than $282k \in$. The maximum and minimum observed MVAs for the three year period ranges from $6,325k \in$ to $-9,664k \in$, respectively.

	MVA	RMVA
Average	-438	-3.70%
Std Deviation	$1,\!838$	32.17%
1st Quartile	-1,111	-16.97%
2nd Quartile	-248	-4.45%
3rd Quartile	282	8.29%
Maximum	6,325	169.10%
Minimum	-9,664	-232.59%

Table 4.4: Global MVA descriptive statistics

Table 4.4 exhibits the descriptive statistics for the global MVA, and respective RMVA, of all companies.

According to Dierks and Patel (1997), Market Value Added, or MVA, is a measure of the wealth a company has created for its investors. Actually, MVA translates the difference between what investors put in and what they can take out of the company. The main feature of the results shown in Table 4.4 was the great contrast in value creation by the Portuguese SME, i.e., more than half of the 315 analyzed companies have destroyed value during 2008-2010.

The global average MVA analysis does not allow us to rank the sectorial performance. In order to do this, we must focus on a sectorial RMVA to rank the most representative sectors of the Portuguese business environment. Table 4.5 reports the sum of all the MVAs and the average the relative MVAs of each sector. Table 4.5 also identifies the MVA and RMVA for the best and worst performers, identifying the respective company ID within each sector. Due to the existence of outliers in our sample, the figures presented in Table 4.5 cannot be directly inferred by the figures presented in Table 4.3.

Between 2008 and 2010, from the eighteen activity sectors, only two contributed positively to the overall MVA. Those sectors were "Electrical Equipment" and "Commercial Services and Supplies", with MVAs of $6,967k \in$ and $715k \in$, respectively. The "Electrical Equipment" sector encompasses the companies producing electrical components for the electronic retailers (mainly retailers for cellphones, smartphones, tablets, computers and other gadgets), whose products have showed consistent growth in revenues from sales. In 2008-2009 companies from this sector have reinforced their positions in the markets in which they operate, experiencing a significant growth in the American, Eastern Europe and Portuguese-speaking African countries.

	Sector			Best Company			Wo	Worst Company		
	MVA	RMVA		MVA	RMVA	ID	MVA	RMVA	ID	
Food Products	-2,553	-30.5%	(6,325	44.0%	8	-4,097	-31.0%	73	
Paper and Forestry Products	-10,888	-89.8%		81	1.7%	89	-4,388	-24.3%	158	
Auto and Components	-6,480	-52.1%	4	$4,\!172$	49.4%	55	-5,183	-28.0%	179	
Electrical Equipment	6,967	71.2%	;	3,412	34.6%	71	-881	-8.1%	171	
Machinery	-5,952	-49.3%	:	$3,\!294$	25.4%	178	-5,710	-32.8%	128	
Pharmaceuticals	-117	-2.0%		1,901	33.2%	12	-2,509	-19.1%	307	
Chemicals	-8,794	-79.8%		$2,\!475$	43.2%	35	-3,921	-16.4%	88	
Textiles, Apparel and Luxury Goods	-15,039	-253.4%		$2,\!689$	59.9%	132	-2,990	-37.4%	187	
Multiline Retailers	-16,037	-385.8%	4	4,788	37.5%	131	-4,272	-59.7%	17	
Specialty Retailers	-37,982	-582.7%		1,206	14.8%	266	-9,569	-29.7%	115	
Electronic Eq. Inst. Comp. Retailers	-617	-11.0%		2,238	36.9%	5	-1,977	-15.1%	9	
Wholesaling	-4,047	-52.0%		2,128	21.1%	127	-4,348	-31.7%	44	
Construction and Engineering	-18,560	-226.1%		2,150	76.3%	269	-9,664	-47.2%	119	
Food/Staple Retailers	-5,395	-159.8%	:	$3,\!307$	75.4%	227	-6,727	-232.6%	301	
Oil and Gas	-2,409	-98.5%		466	63.2%	298	-773	-40.7%	170	
Metals and Mining	-8,041	-92.7%	:	$3,\!252$	65.9%	219	-5,667	-32.9%	108	
Commercial Services and Supplies	715	20.7%		517	25.8%	254	-277	-6.5%	245	
Transportation	-2,800	-43.9%		1,888	35.0%	49	-4,727	-51.6%	90	
Representative company	-438	-5.99%								

Table 4.5: Total MVA, average RMVA, best and worst company (2008-2010)

Table 4.5 aggregates information of different natures. In the second and third columns, MVA is the sum of all the MVAs and RMVA is the average of the relative MVAs of each sector. The remaining six columns report the MVA and RMVA for the best performance ("Best Company") and worst performance ("Worst Company") within each sector, identifying with the respective company ID. The representative company refers to a theoretical company, which portraits as the average SME of our whole sample. The MVA and RMVA for the representative company are computed using average EVAs and average IC, as presented in Table 4.2.

All the other sectors exhibited negative MVAs, with special highlight to the "Specialty Retailers" sector, which registered the lowest MVA of $-37,982k\in$, which is more than the double of the second worst sector, "Construction and Engineering". These sectors were two of the most exposed to the financial crises, in virtue of the decrease in auto sales and the contraction in the construction activity.

Concerning "Specialty Retailers", which corresponds to the commercialization of vehicles, the registered MVA reflects the accentuated decrease in sales for the three year period 2008 2010. These negative results led to an important restructuring process, involving human and capital resources. "Construction and Engineering" sector has been affected by a significant contraction on its production volume for the triennium 2008-2010—registering a cumulated drop of approximately 30% since 2002—mainly influenced by the significantly negative performance of the housing sector. The slight improvement on this sector's performance registered by 2010, is due to the contribution of the international activity—companies from this sector have registered a remarkable growth in revenues concerning foreign activity.

Along with these two sectors, "Multiline Retailers" and "Textile, Apparel and Luxury Goods" recorded the third and fourth worst performances, $-16,037k \in$ and $-15,039k \in$, respectively, which could be explained by the considerable contraction of the private consumption during the crisis period.

4.3. EVA and MVA rankings

This section focuses on the performance analysis to a deeper level, comparing performances of companies rather than sectors. The aim of this analysis is to unravel which companies have created/destroyed more value, establishing a ranking that compares the companies' performance regardless of their sector of activity. Once again, to purge the effect of the different level of IC by each company, the comparison is based on each company's EVA Spread and, afterwards, each company's RMVA.

Table 4.6 and Table 4.7 rank the top ten companies that had the best and the worst performances, respectively. For each company listed in these rankings, we present the EVA Spread for that year, the previous position in the performance ranking, the difference in percentage points of the EVA Spread over the previous year and the sector to which that company belongs. Panel A, B and C refer to the years of 2008, 2009 and 2010, respectively.

EVA TOP+ analysis. As shown in Table 4.6, company number 295 exhibits, for the three years, the best performance, with an EVA Spread of 56.20%, 45.97% and 64.90%, respectively. This company belongs to the "Electronic Equipment, Instruments and Components Retailers" sector.

The ten companies with the best performances of 2008 registered EVA Spreads between 56.20% and 26.10%. The sectors with most representativeness are "Transportation", with two companies, classified in third and fourth places, and "Food/Staple Retailers", also with two companies, in the fifth and eighth places.

Company		Previous	EVA	Δ (p.p.)	
ID	Rank	Year Rank	Spread	(last year)	Sector
Panel A: 20	08				
295	1	n.c.	56.20%	n.c.	Electronic Eq. Inst. Comp. Retailers
269	2	n.c.	53.11%	n.c.	Construction and Engineering
212	3	n.c.	48.93%	n.c.	Transportation
183	4	n.c.	46.94%	n.c.	Transportation
271	5	n.c.	44.91%	n.c.	Food/Staple Retailers
224	6	n.c.	35.38%	n.c.	Oil and Gas
178	7	n.c.	27.26%	n.c.	Machinery
247	8	n.c.	27.01%	n.c.	Food/Staple Retailers
55	9	n.c.	26.32%	n.c.	Auto and Components
206	10	n.c.	26.10%	n.c.	Wholesaling
Panel B: 20	09				
295	1	1	45.97%	-10.23	Electronic Eq. Inst. Comp. Retailers
297	2	20	44.44%	26.33	Multiline Retailers
212	3	3	34.73%	-14.20	Transportation
227	4	26	28.57%	14.60	Food/Staple Retailers
120	5	34	26.90%	16.35	Machinery
298	6	41	24.03%	15.07	Oil and Gas
274	7	12	20.53%	-2.58	Food/Staple Retailers
55	8	9	18.55%	-7.77	Auto and Components
247	9	8	18.46%	-8.56	Food/Staple Retailers
12	10	56	17.36%	11.93	Pharmaceuticals
Panel C: 20	10				
295	1	1	64.90%	18.93	Electronic Eq. Inst. Comp. Retailers
253	2	47	45.06%	38.34	Wholesaling
297	3	2	34.28%	-10.16	Multiline Retailers
227	4	4	32.30%	3.73	Food/Staple Retailers
228	5	123	28.66%	28.92	Food/Staple Retailers
269	6	52	28.27%	22.16	Construction and Engineering
219	7	14	27.38%	11.35	Metals and Mining
132	8	31	27.35%	17.88	Textiles, Apparel and Luxury Goods
52	9	83	26.73%	24.55	Specialty Retailers
299	10	35	26.17%	17.10	Specialty Retailers

Table 4.6: EVA TOP+

Table 4.6 ranks the top ten companies that had the best performances for the years of 2008 (Panel A), 2009 (Panel B) and 2010 (Panel C), in an EVA Spread basis. The first column, "Company ID" identifies the company's ID in the database, column "Rank" displays the raking of the company in the respective year. "Previuous Year Rank" refers to the ranking observed in the previous year. " Δ (p.p.) (last year)" shows the difference, in percentual points, relative to the previous year's EVA Spread. The acronym "n.c." means "not computed".

The best performers of 2009 display EVA Spreads between 45.97% and 17.36%, confirming the generalized decrease in EVA Spread in 2009. "Food/Staple Retailers" is the most represented sector. In 2010, EVA Spread ranges between 64.90% and 26.73%. "Food/Staple Retailers" is once again amongst the most represented companies, along with "Specialty Retailers".

Of the ten companies ranked as the best of 2008, five are present in the second year's ranking. In 2010, only two of the first ten figure in the ranking.

EVA TOP– **analysis.** As presented in Table 4.7, company number 301, from "Food/Staple Retailers", registered the worst performance in the years of 2008 and 2009, with an EVA Spread of -103.58%. In 2010, the company with the worst performance is number 234, from "Food Products", which ranked as third and second in the previous years, respectively.

In 2009, the ten companies with the worst performance registered EVA Spreads between 99.84% and -20.81%. In the last year, this variable oscillates between -60.26% and -18.32%. Even though these companies ranked as the worst performers, there is an improvement, year after year, in their overall performance.

Five of the ten worst companies ranked in 2008 persist in the 2009 ranking, and two of the first ten are listed in the 2010 ranking.

RMVA ranking. Table 4.8 lists the best (Panel A) and worst (Panel B) performing companies for the overall period between 2008 and 2010, in a relative MVA basis (RMVA).

Unsurprisingly, the company that registered the highest RMVA belongs to "Electrical Equipment, Instruments and Components Retailers" sector, identified as company number 295. This company exactly the same that presented the highest EVA Spread for each of the three years and, by 2010, reached a RMVA of 169.10%

By contrast, and also, unsurprisingly, the company with the lowest RMVA is company number 301, of "Food/Staples Retailers". Even though this company was not included in 2010's EVA TOP– ranking, it was the worst performing company for the consecutive years of 2008 and 2009.

Company		Previous	EVA	Δ (n n)	
ID	Rank	Vear Bank	Spread	(last vear)	Sector
Panel A: 20	108	Tear Itank	opread	(last year)	50000
301	315	n.c.	-103.58%	n.c.	Food/Staple Retailers
294	314	n.c.	-55.97%	n.c.	Food/Staple Retailers
234	313	n.c.	-50.64%	n.c.	Food Products
253	312	n.c.	-43.49%	n.c.	Wholesaling
118	311	n.c.	-40.82%	n.c.	Construction and Engineering
255	310	n.c.	-39.27%	n.c.	Food/Staple Retailers
150	309	n.c.	-36.78%	n.c.	Food Products
199	308	n.c.	-34.74%	n.c.	Specialty Retailers
125	307	n.c.	-34.73%	n.c.	Specialty Retailers
80	306	n.c.	-27.83%	n.c.	Construction and Engineering
Panel B: 20	09				- 0 0
301	315	315	-99.84%	3.74%	Food/Staple Retailers
234	314	313	-52.08%	-1.44%	Food Products
255	313	310	-45.53%	-6.26%	Food/Staple Retailers
226	312	58	-37.96%	-43.31%	Pharmaceuticals
199	311	308	-32.80%	1.94%	Specialty Retailers
300	310	37	-31.73%	-41.16%	Food/Staple Retailers
125	309	307	-30.63%	4.10%	Specialty Retailers
272	308	304	-22.18%	4.77%	Textiles, Apparel and Luxury Goods
90	307	264	-21.02%	-9.78%	Transportation
107	306	136	-20.81%	-18.56%	Food Products
Panel C: 20	010				
234	315	314	-60.26%	-8.18%	Food Products
125	314	309	-39.42%	-8.79%	Specialty Retailers
199	313	311	-34.58%	-1.78%	Specialty Retailers
162	312	247	-26.73%	-20.21%	Textiles, Apparel and Luxury Goods
100	311	213	-25.13%	-20.97%	Textiles, Apparel and Luxury Goods
220	310	246	-21.65%	-15.19%	Food/Staple Retailers
119	309	300	-20.94%	-5.08%	Construction and Engineering
80	308	77	-18.89%	-21.71%	Construction and Engineering
17	307	304	-18.77%	-0.58%	Multiline Retailers
288	306	210	-18.32%	-14.32%	Specialty Retailers

Table 4.7: EVA TOP-

Table 4.7 ranks the top ten companies that had the worst performances for the years of 2008, 2009 and 2010, in an EVA Spread basis. The first column, "Company ID" identifies the company's ID in the database, column "Rank" displays the raking of the company in the respective year. "Previuous Year Rank" refers to the ranking observed in the previous year. " Δ (p.p.) (last year)" shows the difference, in percentual points, relative to the previous year's EVA Spread. The acronym "n.c." means "not computed".

The sectors with most representativeness in the MVA Panel A ranking were the "Transportation" and the "Food/Staple" sectors, both with two companies listed in the ranking. As for the MVA Panel B ranking, the sector that stood out was the "Specialty Retailers", with three companies listed in the ranking.

Company ID	RMVA	Sector
Panel A: Best	performers	
295	169.10%	Electronic Eq. Inst. Comp. Retailers
297	99.18%	Multiline Retailers
212	91.93%	Transportation
269	76.31%	Construction and Engineering
227	75.38%	Food/Staple Retailers
183	72.63%	Transportation
219	65.88%	Metals and Mining
298	63.18%	Oil and Gas
271	62.20%	Food/Staple Retailers
132	59.87%	Textiles, Apparel and Luxury Goods
Panel B: Worst	performers	
301	-232.59%	Food/Staple Retailers
234	-164.99%	Food Products
125	-104.59%	Specialty Retailers
199	-102.03%	Specialty Retailers
255	-93.89%	Food/Staple Retailers
17	-59.71%	Multiline Retailers
150	-52.69%	Food Products
90	-51.58%	Transportation
14	-49.84%	Specialty Retailers
119	-47.21%	Construction and Engineering

Table 4.8:TOP RMVA

Table 4.8 lists the best (Panel A) and worst (Panel B) performing companies for the overall period between 2008 and 2010, in a relative MVA basis (RMVA).

Companies from "Food/Staple Retailers" sector appear in the EVA TOP+, EVA TOPand both Panel A and B of the MVA TOP rankings. The explanation for this phenomenon could be in the fact that "Food/Staple Retailers" is a sector with many players of considerably different sizes and talents. The biggest players, which are few in number, dominate the market, pushing the many small players to the bottom of the ranking.

CHAPTER 5

CONCLUDING REMARKS

Over the last two decades, the rapid evolution of technology and globalization have brought new and more demanding challenges to today's managers and business executives. These developments bring changes to the economic reality of each country, to which all companies must adapt. By doing so, their performance may suffer and must be measured in a continuous basis and controlled in order to guide their strategic management.

As a direct consequence of the importance of performance measurement and control, value based performance measures, such as shareholder value added, economic value added, economic profit and cash flow return on investment have gained increased popularity in recent years.

In this context, traditional accounting measures, used individually, cannot satisfy the demands of today's economies. EVA stands out from traditional accounting measures since it can be used as a multidimensional single tool that may answer almost all of today's market demands—it can be used to measure value, to measure performance, as a management, decision-making and incentive compensation tool. However, some results of the empirical research to date are not totally consistent with those statements. They are, in fact, mixed and controversial.

This dissertation is driven by the attempt to understand if the Portuguese SME have been able to keep up with the market developments in the years 2008, 2009 and 2010, and still managed to create value. As far as we know, this is the first work which attempts to calculate and assess the performance of the Portuguese SME in terms of economic value added and their contribute the Portuguese economy.

Several selection criteria were applied to a raw database of 4,000 companies, with the purpose of obtaining a sufficiently homogeneous sample. The result was a total sample of 315 companies, from 18 different sectors, that represent the biggest and best Portuguese SME for the years of 2008, 2009 and 2010. We computed EVA for each of these companies, with the purpose of determining the value added by each company for the three year period and then analyzed the results.

Globally, the total EVA is negative in each of the three years, adding a total of $-79,769k \in$ in 2008, $-49,824k \in$ in 2009 and $-8,437k \in$ in 2010. However, the number of companies with negative EVA decreases by each year, from 216 companies in 2008, to 197 companies in 2009, to 157 companies in 2010. The total MVA is negative and adds up to $-138,029k \in$, influenced by the 200 companies that contribute with negative MVAs.

The results of the assessment performance in this dissertation lead us to the conclusion that, for the period of 2008-2010, the biggest and best Portuguese SME have not created value, as reflected by the total MVA. The comparison of financing conditions to the returns on the companies' activities translates in a negative EVA Spread for the first two years and positive for the last, i.e., on average terms, the EVA Spread was -2.84% in 2008, -1.13% in 2009 and 2.67% in 2010. These numbers are the reflex of the uncompetitive conditions of funding, especially in the short-term, as a result of the companies' chronic lack of liquidity. However, as noted especially in the year 2010, this panorama seems to be improving for the Portuguese SME.

The years of 2008, 2009 and 2010 were profoundly affected by the outbreak of the economic and financial international crisis, which could explain the bulk of these negative results. Nevertheless, other factors may have influenced negatively the performance of the Portuguese SME. Among these factors, we can consider the low level of internationalization outside Europe where the effects of the crisis were very strong—, the strong impact on the Portuguese economy derived from the social context—which has reduced significantly the private consumption levels—the lack of modernization and growth, together with the lack of entrepreneurial culture experienced in Portugal, and the low levels of technological orientation of the main chain of value of our traditional sectors, prevent the Portuguese businesses from competing globally with companies from other emerging countries such as China, India and Brazil. One of the main difficulties in the course of this dissertation lies in the lack of publicly available information on the Portuguese sectors and companies' systematic risk, which forced us to use estimates, eventually causing distortions from the economic reality we aim to portray. The second constraint is related to the fact that we could not obtain data on the financial statements of these companies for the years of 2011 and 2012, limiting the robustness of the results of our performance assessment. Finally, since this is, as far as we know, the first empirical work on the Portuguese SME performance measurement, there were no other means of comparing our results with previous works.

In terms of future research, efforts should be made in order to assess the combined impact from two major factors that have affected the Portuguese economy after 2010. Those factors are the generalized effort employed in raising exports of products and services provided by Portuguese companies and the economic and financial intervention, from the International Monetary Fund and The European Financial Stability Facility, inducing discipline and rigor by the implementation of austerity measures, fiscal consolidation process and the private sector deleveraging. Also, we believe that, in the future, efforts should be made in order to develop a mechanism to compute and publish, on a yearly basis, the performance of Portuguese SME as measured by economic value added. This procedure would allow the Portuguese authorities to better identify and stimulate the sectors which present the best performances and contribute the most to the generation of wealth.

References

- Abdeen, A. and G. Haight, 2002, A Fresh Look at Economic Value Added: Empirical Study of the Fortune Five-Hundred Companies, *Journal of Applied Business Research* 18, 27–35.
- Athanassakos, G., 2007, Value-Based Management, EVA and Stock Price Performance in Canada, *Management Decision* 45, 1397–1411.
- Bahri, M., J. St-Pierre, and O. Sakka, 2011, Economic Value Added: A Useful Tool for SME Performance Measurement, International Journal of Productivity and Performance Management 60, 603–621.
- Banco de Portugal, 2013, The Portuguese Economy in 2012, Banco de Portugal: Annual Report, 1–136.
- Bao, B-H. and D-H. Bao, 1998, Usefulness of Value Added and Abnormal Economic Earnings: An Empirical Examination, *Journal of Business Finance and Accounting* 25, 251–264.
- Biddle, G., R. Bowen, and J. Wallace, 1997, Does EVA Beat Earnings? Evidence on Associations with Stock Returns and Firm Values, *Journal of Accounting and Economics* 24, 301–336.
- Binnersley, M., 1996, Do You Measure Up?, Management Accounting 74, 32–34.
- Chen, S. and J. Dodd, 1997, Economic Value Added (EVA): An Empirical Examination of a New Corporate Performance Measure, *Journal of Managerial Issues* 9, 318–333.
- Clinton, B. and S. Chen, 1998, Do New Performance Measures Measure Up?, *Management Accounting* 80, 38–44.
- Cocca, P. and M. Alberti, 2010, A Framework to Assess Performance Measurement Systems in SMEs, International Journal of Productivity and Performance Management 59, 186–200.
- Cordeiro, J. and D. Kent, 2001, Do EVA Adopters Outperform their Industry Peers? Evidence from Security Analyst Earning Forecasts, American Business Review 19, 57–63.
- Correia, C., D. Flynn, E. Uliana, and M. Wormald, 2007, *Financial Management*, 6th ed. (Juta and Company, Cape Town).
- Damodaran, A., 2001, Corporate Finance: Theory and Practice, Wiley International Edition, 2nd ed. (John Wiley and Sons, New York).
- DeWet, J., 2005, EVA versus Traditional Accounting Measures of Performance Drivers of Shareholder Value - A Comparitive Analysis, *Meditari Accountancy Research* 13, 1–16.

- Dierks, P. and A. Patel, 1997, What is EVA, and How Can It Help Your Company?, Management Accounting 79, 52–58.
- Diário Económico, 2010, As 5000 PME Que Lideram em Portugal, *Revista Integrada no DE No. 4827*, February.
- Dow Theory Forecast, 1999, Companies with Confidence, Horizon Publishing Company 55.
- Drury, C., 2007, *Management and Cost Accounting*, 7th ed. (South-Western Publishers, Massachussets).
- Elali, W., 2007, EVA and Shareholder Value Creation: An Empirical Study, Westminster Business School, PhD Dissertation.
- Erasmus, P., 2008, The Relative Incremental Information Content of the Value Based and Financial Performance Measure Cash Value Added (CVA), *Management Dynamics* 17, 2–15.
- Eurostat, 2011, Taxation Trends in the European Union, Eurostat and European Comission Tech Report.
- Fernandez, P., 2001, EVA Economic Profit and Cash Value Added do Not Measure Shareholder Value Creation, Journal of Applied Finance 9, 74–94.
- Fernandez, P. and J. Del Campo, 2010, Survey MRP Used in 2010 by Professors: 1,500 Answers, IESE Business School, Working Paper.
- Fernandez, P., J. Aguirreamalloa, and P. Linares, 2013, Market Risk Premium and Free Risk Rate Used for 51 Countries in 2013: A Survey with 6,237 Answers, *IESE Business School*, *Working Paper*.
- Finegan, Y., 1991, Maximizing Shareholder Value at the Private Company, Journal of Corporate Applied Finance 4, 30–45.
- Foltin, C., 1999, State and Local Government Performance: It's Time to Measure Up!, Government Accountants Journal 48, 40–46.
- Frezati, F., 2003, Gestão de Valor Na Empresa: Uma Abordagem Abrangente do Valuation a Partir da Contabilidade Gerencial, 1st ed. (Atlas, São Paulo).
- Ghanbari, M. and V. More, 2007, The Relationship Between Economic Value Added and Market Value Added: An Empirical Analysis in Indian Automobile Industry, *The ICFAI Journal of Accounting Research* 6, 7–22.
- Godfrey, S. and R. Espinosa, 1996, A Practical Approach to Calculating Costs of Equity for Investments in Emerging Markets, *Journal of Applied Corporate Finance* 9, 80–90.
- Grant, J., 1996, Foundations of EVA for Investment Managers, Journal of Portfolio Management 23, 41–48.

- Hamada, R., 1972, The Effect of the Firm's Capital Structure on the Systematic Risk of Common Stocks, *Journal of Finance* 27, 435–452.
- Herzberg, M., 1998, Implementing EBO/EVA Analysis in Stock Selection, Journal of Investing, Spring, 45–53.
- Hussein, M. and E. Laitinen, 1998, Management Accounting System in Finnish Service Firms, *Technovation* 18, 57–67.
- Instituto Nacional de Estatística, 2010, Estudos Sobre Estatísticas Estruturais das Empresas, destaque, June.
- Ismail, Issham, 2013, Economic Value Added (EVA) versus Traditional Tools in Predicting Corporate Performance in Malaysia, African Journal of Business Management 7, 1757– 1764.
- Jensen, M. and K. Murphy, 1990, CEO Incentives: It's Not How Much You Pay, But How?, Harvard Business Review 3, 138–153.
- Joshi, S., 2011, Relationship Between EVA, MVA and Other Accounting Measures of Fertilizer Companies in India, International Journal of Research in IT and Management 1, 1–14.
- Kleiman, R., 1999, Some New Evidence on EVA Companies, Journal of Applied Corporate Finance 12, 80–91.
- Kramer, J. and G. Pushner, 1997, An Empirical Analysis of Economic Value Added as a Proxy for Value Added, *Financial Practice and Education* 7, 41–49.
- Lehn, K.; and A. Makhija, 1997, EVA, Accounting Profits and CEO Turnover: An Empirical Explanation, *Journal of Applied Corporate Finance* 10, 90–96.
- Lovata, L. and M. Costigan, 2002, Empirical Analysis of Adopters of Economic Value Added, Management Accounting Research 13, 215–228.
- McConville, D., 1994, All About EVA, Industry Week 243, 55–58.
- McCormack, J. and J. Vytheeswaran, 1998, How to Use EVA in the Oil and Gas Industry, Journal of Applied Corporate Finance 11, 109–132.
- O'Byrne, S., 1996, EVA and Market Value, Journal of Applied Corporate Finance 9, 116–125.
- OECD, 2009, The Impact of the Global Crisis on SME and Entrepreneurship Financing and Policy Responses, Organisation for Economic Co-operation and Development - Centre for Entrepreneurship, SMEs and Local Development, 1–69.
- Peixoto, S., 2001, Economic Value Added Aplication to Portuguese Public Companies, Modern University of Porto, MSc Dissertation.
- Raiyani, J. and N. Joshi, 2011, EVA Based Performance Measurement: A Case Study of SBI and HDFC Bank, *Management Insight* 7, 31–43.

- Ray, R., 2001, Economic Value Added: Theory, Evidence, a Missing Link, *Review of Business* 22, 66–70.
- Raymond, L., J. St-Pierre, and M. Marchand, 2009, A Taxonomic Approach to Studying the Performance of Manufactoring SMEs, *International Journal of Business Performance Management* 11, 277–291.
- Roztocki, N. and K. Needy, 1999, EVA for Small Manufactoring Companies, Proceedings from the 1999 SAM International Management Conference, March, 461–469.
- Sharma, A. and S. Kumar, 2010, Economic Value Added (EVA) Literature Review and Relevant Issues, *International Journal of Economics and Finance* 2, 200–220.
- Singh, T. and S. Mehta, 2012, EVA vs Traditional Accounting Measures: A Pre Recession Case Study of Selected IT Companies, International Journal of Marketing and Technology 2, 95–120.
- Stern, J., G. Stewart, and D. Chew, 1995, The EVA Financial Management System, Journal of Applied Corporate Finance 8, 32–46.
- Stern, J., G. Stewart, and D. Chew, 1998, The Revolution in Corporate Finance, 3rd ed. (Blackwell Publishers, Oxford).
- Stewart, G., 1991, *The Quest for Value: A Guide for Senior Managers* (Harper Business, New York).
- Stewart, G., 1994, EVA: Fact and Fantasy, Journal of Applied Corporate Finance 7, 71-84.
- Stewart III, G. and A. Ehrbar, 1999, The EVA Revolution, Journal of Applied Corporate Finance 12, 18–31.
- Tham, J., 2001, The Unbearable Lightness of EVA in Valuation, *Duke University Duke Center for International Development in the Sanford School of Public Policy*. Available at SSRN: http://ssrn.com/abstract=267193.
- Uyemura, D., C. Kantor, and J. Petit, 1996, EVA for Banks: Value Creation, Risk Management and Profitability Measurement, *Journal of Applied Corporate Finance* 9, 94–111.
- Van der Poll, H., N. Booyse, A. Pienaar, S. Büchner, and J. Foot, 2011, An Overview of the Implementation of Economic Value Added (EVA) Performance Measures in South Africa, *Southern African Business Review* 15, 122–140.
- Wallace, J., 1994, Adopting Residual Income-Based Compensation Plans: Do You Get What You Pay for?, Journal of Accounting and Economics 24, 275–300.
- Weaver, S., 2001, Measuring Economic Value Added: A Survey of the Practices of EVA Proponents, *Journal of Applied Finance* 11, 50–60.
- Worthington, A. and T. West, 2004, Australian Evidence Concerning the Information Content of Economic Value Added, *Australian Journal of Management* 29, 201–224.

Young, D., 1999, Some Reflections on Accounting Adjustments and Economic Value Added, Journal of Financial Statement Analysis 4, 7–13.

APPENDIX A

Annexes

Table A.1: Sector names and Portuguese correspondents

Sector name (English)	Correspondent sector name (Portuguese)
Food Products	Agricultura e Agro-indústria
Paper and Forestry Products	Celulose e papel, madeira, cortiça e móveis
Auto and Components	Industria Automóvel
Electrical Equipment	Material eléctrico e de precisão
Machinery	Metalomecânica e metalurgia de base
Pharmaceuticals	Produtos farmacêuticos
Chemicals	Química
Textiles, Apparel and Luxury Goods	Texteis, Vestuário e Couro
Multiline Retailers	Comércio a Retalho
Specialty Retailers	Comércio de veículos automóveis
Electronic Eq. Inst. and Comp. Retailers	Comércio electro-electrónico
Wholesaling	Comércio por grosso
Construction and Engineering	Construção
Food/Staple Retailers	Distribuição alimentar
Oil and Gas	Distribuição de combustíveis
Metals and Mining	Minerais metálicos e não metálicos
Commercial Services and Supplies	Serviços
Transportation	Transportes e distribuição

This table presents each sector's name in Portuguese, according to the codes in "Classificação Portuguesa das Actividades Económicas—CAE Rev. 3, Ano 2007" and its proxy in English, according to the MSCI Global and Sectorial Indexes in "GICS (Global Industry Classification Standard)—from Standard & Poors and MSCI Barra, Year 2010".

Table A.2: List of abbreviations and acronyms

CEO	Chief Executive Officer
EBIT	Earnings Before Interest and Taxes
\mathbf{EPS}	Earnings Per Share
EUR	Euro Currency
EVA	Economic Value Added
GAAP	Generally Accepted Accounting Principles
IAPMEI	Instituto de Apoio às Pequenas e Médias Empresas
IC	Invested Capital
ID	Identification number in the database
IESE	Instituto de Estudos Superiores da Empresa
INE	Instituto Nacional de Estatística
MSCI	Morgan Stanley Capital International
MVA	Market Value Added
NI	Net Income
NOPAT	Net Operating Profits After Taxes
OECD	Organization for the Economic Cooperation and Development
OP	Operating Profit
PME	Pequenas e Médias Empresas
RMVA	Relative Market Value Added
ROE	Return on Equity
ROIC	Return on Invested Capital
SME	Small and Medium Enterprises
k€ M€ b.p. p.p.	Thousands of Euros Millions of Euros Basis points Percentual points